SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM 8-K

CURRENT REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

June 18, 2008

Date of report (Date of earliest event reported)

DELUXE CORPORATION

(Exact Name of Registrant as Specified in its Charter)

Minnesota	1-7945	41-0216800		
(State of Incorporation)	(Commission File Number)	(I.R.S. Employer		
		Identification No.)		
3680 Victoria St. N.				
Shoreview, Minnesota		55126		
(Address of Principal Executive Offices)		(Zip Code)		
	((51) 402 5111			
	(651) 483-7111			
	(Registrant's Telephone Number, Including Area Code)			
Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions (see				
General Instruction A.2):				
o Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)				
o Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)				
o Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))				
o Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))				

TABLE OF CONTENTS

Item 1.01. Entry into a Material Definitive Agreement
Item 9.01 Financial Statements and Exhibits
SIGNATURES
EXHIBIT INDEX
Agreement and Plan of Merger
Voting Agreement
Joint Press Release

Item 1.01. Entry into a Material Definitive Agreement

On June 18, 2008, Deluxe Corporation ("Deluxe"), Deluxe Business Operations, Inc., a wholly owned subsidiary of Deluxe ("Operating Sub"), Helix Merger Corp., a wholly owned subsidiary of Operating Sub ("Merger Sub"), and Hostopia.com Inc. ("Hostopia") entered into an Agreement and Plan of Merger (the "Merger Agreement") pursuant to which, among other things, Merger Sub will merge with and into Hostopia (the "Merger"). Subject to the terms and conditions of the Merger Agreement, which has been unanimously approved by the Boards of Directors of both Deluxe and Hostopia, each share of Hostopia common stock will be converted into the right to receive C\$10.55 in cash, without interest and subject to any applicable tax withholdings, if the Merger is completed.

The Merger Agreement contains representations, warranties and covenants of Deluxe and Hostopia. The board of directors of Hostopia has adopted a resolution recommending the approval and adoption of the Merger Agreement by Hostopia's stockholders, and has agreed to cause a stockholder meeting to be held to consider the Merger Agreement. In addition, Hostopia has agreed not to (i) solicit proposals relating to alternative business combination transactions or (ii) subject to certain exceptions, enter into discussions or negotiations concerning, or provide confidential information in connection with, alternative business combination transactions.

Consummation of the Merger, which is currently anticipated to occur in the third quarter of 2008, is subject to customary closing conditions, including approval by Hostopia's stockholders. In connection with the Merger Agreement, certain stockholders of Hostopia entered into a voting agreement with Deluxe (the "Voting Agreement") pursuant to which such stockholders agreed to vote in favor of the approval and adoption of the Merger Agreement and the Merger and against any alternative business combination transactions, as long as the Merger Agreement has not been terminated.

The Merger Agreement contains certain termination rights for both Deluxe and Hostopia, and further provides that, upon termination of the Merger Agreement under specified circumstances, Hostopia may be required to pay Deluxe a termination fee and expense reimbursement of up to C\$4.8 million in the aggregate.

The foregoing summaries of the Merger Agreement and the Voting Agreement do not purport to be complete and are qualified in their entirety by reference to the Merger Agreement, which is attached as Exhibit 2.1 hereto and is incorporated herein by reference, and the Voting Agreement, which is attached as Exhibit 99.1 hereto and is incorporated herein by reference. A copy of the press release announcing the execution of the Merger Agreement is also attached to this report as Exhibit 99.2.

Item 9.01 Financial Statements and Exhibits

(d) Exhibits

2.1 Agreement and Plan of Merger, dated as of June 18, 2008, among Deluxe Corporation, Deluxe Business Operations, Inc., Helix Merger Corp. and

- Hostopia.com Inc. (excluding schedules, which Deluxe Corporation agrees to furnish to the Securities and Exchange Commission upon request)
- 99.1 Voting Agreement, dated as of June 18, 2008, among Deluxe Corporation, William Campbell, Colin Campbell, Michael Cytrynbaum, Mathew George, Robert Kidd, Randall Bast, Christopher Scatliff, 1999 William Campbell Family Trust, 1555706 Ontario Limited and TELUS Corporation
- 99.2 Joint Press Release of Deluxe Corporation and Hostopia.com Inc., dated June 19, 2008, announcing the signing of a definitive agreement

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

DELUXE CORPORATION

Date: June 23, 2008 /s/ Anthony Scarfone

Name: Anthony Scarfone

Title: Senior Vice President, General Counsel and Secretary

EXHIBIT INDEX

No.	Description
2.1	Agreement and Plan of Merger, dated as of June 18, 2008, among Deluxe Corporation, Deluxe Business Operations, Inc., Helix Merger Corp. and Hostopia.com Inc.
99.1	Voting Agreement, dated as of June 18, 2008, among Deluxe Corporation, William Campbell, Colin Campbell, Michael Cytrynbaum, Mathew George, Robert Kidd, Randall Bast, Christopher Scatliff, 1999 William Campbell Family Trust, 1555706 Ontario Limited and TELUS Corporation
99.2	Joint Press Release of Deluxe Corporation and Hostopia.com Inc., dated June 19, 2008, announcing the signing of a definitive agreement

AGREEMENT AND PLAN OF MERGER

among

Deluxe Corporation,

Deluxe Business Operations, Inc.,

Helix Merger Corp.,

and

Hostopia.com Inc.

Dated as of June 18, 2008

TABLE OF CONTENTS

ARTICLE I THE MERGER	1
1.1 Generally	1
1.2 Effective Time	2
1.3 Effects	2
1.4 Organizational Documents of the Surviving Corporation	2
1.5 Directors and Officers of the Surviving Corporation	2
1.6 Conversion of Shares	2
1.7 Dissenters' Rights	3
1.8 Stock Options	4
1.9 Payment for Shares	5
1.10 No Further Rights or Transfers	6
ARTICLE II CLOSING	7
2.1 Time and Place	7
2.2 Deliveries at the Closing	7
ARTICLE III REPRESENTATIONS AND WARRANTIES OF THE COMPANY	7
3.1 Corporate Organization and Qualification	7
3.2 Capitalization	8
3.3 Authority; Non-Contravention	10
3.4 Securities Reports	11
3.5 Undisclosed Liabilities	13
3.6 Absence of Changes	13
3.7 Taxes	13
3.8 Properties	15
3.9 Contracts	16
3.10 Intellectual Property	18
3.11 Litigation	22
3.12 Compliance with Laws	22
3.13 Licenses and Permits	23
3.14 Employee Plans	23
3.15 Employment and Labor Matters	26
3.16 Environmental	26
3.17 Suppliers and Customers	27
3.18 Insurance	28
3.19 Competition Act (Canada)	28
3.20 Antitakeover Provisions	28
3.21 Stockholder Voting Requirement	28
3.22 Associate Transactions	28
3.23 Unregistered Sale of Securities	29

i

3.24 Brokers; Finders; Transaction Fees 3.25 Fairness Opinion	29 29
ARTICLE IV REPRESENTATIONS AND WARRANTIES OF PARENT, OPERATING SUB, AND SUB 4.1 Corporate Organization 4.2 Authority; Non-Contravention 4.3 Adequate Funds 4.4 Litigation 4.5 Finders; Brokers	29 29 30 30 30 31
ARTICLE V PRE-CLOSING COVENANTS 5.1 Operation of Business of the Company 5.2 Stockholders' Meeting; Information Circular 5.3 No Solicitation 5.4 Access to Information 5.5 Employee Benefit Plans 5.6 Reasonable Best Efforts 5.7 Consents 5.8 Public Announcements 5.8 Public Announcements 5.9 Notification of Certain Matters 5.10 Approval of Sub's Sole Stockholder	31 31 34 35 39 39 41 41 41
ARTICLE VI OTHER COVENANTS 6.1 Indemnification 6.2 D&O Liability Insurance	41 41 42
ARTICLE VII CONDITIONS 7.1 Conditions to the Obligations of Parent, Operating Sub, and Sub 7.2 Conditions to the Obligation of the Company ARTICLE VIII TERMINATION	43 43 45
8.1 Termination 8.2 Procedure and Effect of Termination 8.3 Termination Fee	46 46 47 48
ARTICLE IX MISCELLANEOUS 9.1 Termination of Representations and Warranties 9.2 Amendment and Modification 9.3 Waiver of Compliance; Consents 9.4 Expenses 9.5 Additional Agreements 9.6 Notices	49 49 49 49 49 49

9.7 Assignment; No Third-Party Beneficiaries	50
9.8 Interpretation	51
9.9 Governing Law	52
9.10 Counterparts	52
9.11 Headings	52
9.12 Disclosure Schedule	52
9.13 Specific Performance	53
9.14 Entire Agreement	53
iii	

AGREEMENT AND PLAN OF MERGER

This Agreement and Plan of Merger (this "<u>Agreement</u>") is made as of June 18, 2008 among Deluxe Corporation, a Minnesota corporation ("<u>Parent</u>"), Deluxe Business Operations, Inc., a Delaware corporation and a wholly owned subsidiary of Parent ("<u>Operating Sub</u>"), Helix Merger Corp., a Delaware corporation and wholly owned subsidiary of Operating Sub ("<u>Sub</u>"), and Hostopia.com Inc., a Delaware corporation (the "<u>Company</u>"). The Company and Sub are sometimes collectively referred to as the "Constituent Corporations."

Recitals

- A. The respective Boards of Directors of the Company, Parent, Operating Sub, and Sub deem a merger of Sub with and into the Company under the terms hereof (the "Merger") advisable, fair to, and in the best interests of their respective corporations and their respective stockholders.
- B. The Boards of Directors of the Company, Parent, Operating Sub, and Sub have, by resolutions duly adopted, unanimously approved this Agreement and the Merger, and the Boards of Directors of the Company and Sub have unanimously directed that this Agreement and the Merger be submitted to a vote of the stockholders of their respective Constituent Corporations in accordance with the laws of the State of Delaware.
- C. Simultaneously with the execution of this Agreement, certain stockholders of the Company have entered into a Voting Agreement with Parent, dated as of the date hereof (the "<u>Voting Agreement</u>"), under which such Company stockholders have, among other things, granted to representatives of Parent irrevocable proxies (the "<u>Irrevocable Proxies</u>") to vote all shares of Company Common Stock (as defined in Section 1.6(a)) held of record by such stockholders to approve this Agreement and the Merger, unless this Agreement shall have been earlier terminated under Section 8.1.
- D. The Company, Sub, Operating Sub, and Parent desire to effect the Merger and the other transactions contemplated hereby on the terms and subject to the conditions set forth herein.

The parties therefore agree as follows:

Article I The Merger

1.1 Generally. At the Effective Time (as defined in Section 1.2) and in accordance with the terms of this Agreement and the General Corporation Law of the State of Delaware (the "Delaware Law"), Sub will be merged with and into the Company, the separate corporate existence of Sub will thereupon cease, and the Company will be the surviving corporation in the Merger (the "Surviving Corporation").

- 1.2 Effective Time. Subject to and on the same business day as, and promptly following, the receipt of the vote of the stockholders of the Company adopting and approving this Agreement and the Merger and the satisfaction or waiver of all other conditions to the consummation of the Merger set forth in Article VII, other than those conditions that by their nature are to be satisfied at the Closing (as defined in Section 2.1), but subject to the satisfaction or waiver of such conditions (unless another time or date is agreed to by Parent and the Company), the Company shall execute in the manner required by the Delaware Law and deliver for filing to the Secretary of State of the State of Delaware a certificate of merger with respect to the Merger in such form as is required by, and executed in accordance with, the relevant provisions of the Delaware Law (the "Certificate of Merger"). The Merger will become effective at such later time and date as Parent and the Company mutually agree and set forth in the Certificate of Merger, which date will be the fourth business day after the Closing Date (as defined in Section 2.1), unless Parent and the Company agree to a different date. The term "Effective Time" means the date and time when the Merger becomes effective.
 - **1.3** Effects. The Merger will have the effects set forth in Section 259 of the Delaware Law.

1.4 Organizational Documents of the Surviving Corporation.

- (a) Subject to Section 6.1, at the Effective Time, the Certificate of Incorporation of the Company will be amended and restated in its entirety to read as set forth on Exhibit A hereto, and, as so amended, will be the Certificate of Incorporation of the Surviving Corporation until amended in accordance with the Delaware Law.
- (b) Subject to Section 6.1, the Bylaws of Sub in effect immediately before the Effective Time will be deemed, by virtue of the Merger and without further action by the stockholders or directors of the Surviving Corporation or Sub, to be the Bylaws of the Surviving Corporation, until further amended in accordance with the Delaware Law and the Surviving Corporation's Certificate of Incorporation and Bylaws, except that all references therein to the name of Sub shall be amended to be references to "Hostopia.com Inc."
- 1.5 <u>Directors and Officers of the Surviving Corporation</u>. The directors of Sub immediately before the Effective Time will be the directors of the Surviving Corporation, each of such directors to hold office, subject to the applicable provisions of the Bylaws of the Surviving Corporation, until the expiration of the term for which such director was elected and until his or her successor is duly elected and qualified or as otherwise provided in the Bylaws of the Surviving Corporation. The officers of Sub immediately before the Effective Time will be the officers of the Surviving Corporation until their respective successors are chosen and qualified or as otherwise provided in the Bylaws of the Surviving Corporation.
 - 1.6 Conversion of Shares. The manner and basis of converting the shares of each of the Constituent Corporations in the Merger at the Effective Time will be as follows:

- (a) Each share of common stock of the Company, par value US\$.0001 per share (the "Company Common Stock"), that is issued and outstanding immediately before the Effective Time (other than (i) Dissenting Shares (as defined in Section 1.7) and (ii) shares of Company Common Stock held of record by Parent, Operating Sub, Sub, or any other direct or indirect subsidiary of Parent or the Company immediately before the Effective Time) will, by virtue of the Merger and without any action on the part of the holder thereof, be converted into and represent the right to receive C\$10.55 in cash (the "Merger Consideration"). If, between the date of this Agreement and the Effective Time, the outstanding shares of Company Common Stock are changed into, or exchanged for, a different number of shares or a different class, by reason of any stock dividend, subdivision, reclassification, reorganization, recapitalization, split, combination, contribution, or exchange of shares, the Merger Consideration will be correspondingly adjusted.
- (b) Each share of common stock of Sub, par value US\$.001 per share, that is issued and outstanding immediately before the Effective Time will, by virtue of the Merger and without any action on the part of the holder thereof, be converted into and exchanged for one share of common stock of the Surviving Corporation, which will constitute the only issued and outstanding shares of capital stock of the Surviving Corporation immediately after the Effective Time.
- (c) Each share of Company Common Stock held of record by Parent, Operating Sub, Sub, or any other direct or indirect subsidiary of Parent or the Company immediately before the Effective Time, and each share of Company Common Stock held in the treasury of the Company immediately before the Effective Time, will be canceled and cease to exist at and after the Effective Time, and no payment will be made with respect thereto.

1.7 Dissenters' Rights.

- (a) Notwithstanding any provision of this Agreement to the contrary, any shares of Company Common Stock outstanding immediately before the Effective Time held by a holder who has properly demanded and perfected the right, if any, for appraisal of those shares in accordance with the provisions of Section 262 of the Delaware Law and as of the Effective Time has not withdrawn or lost such right to such appraisal ("Dissenting Shares") shall not be converted into or represent a right to receive the Merger Consideration, and the holder thereof shall only be entitled to payment of the fair value of such shares in accordance with the provisions of Section 262 of the Delaware Law. At the Effective Time, the Dissenting Shares will no longer be outstanding and will automatically be canceled and will cease to exist, and each holder of Dissenting Shares will cease to have any rights with respect thereto, except the right to receive the fair value of such shares in accordance with the provisions of Section 262 of the Delaware Law.
 - (b) If a holder of shares of Company Common Stock who demands appraisal of those shares under the Delaware Law effectively withdraws or loses (through failure to

perfect or otherwise) the right to appraisal, then those shares will be deemed to have been converted at the Effective Time into and represent only the right to receive the Merger Consideration as provided in Section 1.6(a), without interest, upon compliance with the provisions, and subject to the limitations, of Section 1.9.

(c) The Company shall give Parent (i) prompt notice of any written demands for appraisal of any shares of Company Common Stock, attempted withdrawals of such demands, and any other instruments received by the Company relating to stockholders' rights of appraisal, and (ii) the opportunity to direct all negotiations and proceedings with respect to demands for appraisal under the Delaware Law. The Company shall not, except with the prior written consent of Parent, voluntarily make any payment with respect to any demands for appraisal of Company Common Stock, offer to settle or settle any such demands, approve any withdrawal of any such demands, or agree or commit to do any of the foregoing.

1.8 Stock Options.

- (a) With respect to any option granted under any employee or director stock option or compensation plan or other arrangement with the Company to purchase shares of Company Common Stock (an "Option") that is exercisable immediately prior to the Closing Date (a "Vested Option"), the Company shall, prior to the Closing Date, take all actions that are necessary to cause the holder of such Vested Option to receive, in cancellation of the Option and in lieu of any other awards with respect to the Option, promptly after the Effective Time, a cash payment from the Surviving Corporation in an amount equal to the amount, if any, by which the Merger Consideration exceeds the per-share exercise price of the Option, multiplied by the number of shares of Company Common Stock subject to the Option immediately before the Effective Time (the "Option Settlement Amount"), subject to the holder's consent, if required, and subject to all required tax withholdings by the Surviving Corporation. Parent shall cause the Option Settlement Amount payable under this Section 1.8 to be paid by the Surviving Corporation as soon as reasonably practicable following the Effective Time but in any event within 15 days thereof.
- (b) With respect to any Option or portion thereof that is not exercisable immediately prior to the Closing Date (an "<u>Unvested Option</u>"), the Company shall, at the direction of Parent, either (i) take such action as may be necessary to convert such Unvested Option into a Vested Option to be settled as provided in paragraph (a) above, (ii) provide for the cancellation of such Unvested Option, subject to the holder's consent, or (iii) take such other action as may be deemed appropriate by Parent, subject to the holder's consent, if required. Parent may direct different treatment for different Option holders.

1.9 Payment for Shares.

(a) On the Closing Date, Parent, Operating Sub, or Sub shall deposit in immediately available funds with Wells Fargo Bank, N.A. or another disbursing agent

selected by Parent that is organized under the laws of the United States or Canada or any state of the United States or province of Canada with capital, surplus, and undivided profits of at least US\$1 billion or C\$1 billion (the "Disbursing Agent"), an amount equal to the product of (i) the number of shares of Company Common Stock issued and outstanding immediately before the Effective Time (other than shares then held of record by Parent, Operating Sub, Sub, or any other direct or indirect subsidiary of Parent or the Company), times (ii) the Merger Consideration (such product referred to as the "Fund"). Out of the Fund, the Disbursing Agent shall make the payments referred to in Section 1.6(a), subject to the requirements of Section 1.9(b). At the request of the Surviving Corporation, in its sole discretion at any time, but without any obligation to make any such request, the Disbursing Agent also may make payments, in discharge of any obligations of the Surviving Corporation under Section 262 of the Delaware Law, to holders of Dissenting Shares. The Disbursing Agent shall invest portions of the Fund as Parent or the Surviving Corporation directs, provided that all such investments shall be held as cash or in obligations of or guaranteed by the United States of America or Canada, or in certificates of deposit, bank repurchase agreements, or bankers' acceptances of commercial banks with capital, surplus, and undivided profits exceeding US\$1 billion or C\$1 billion (collectively, "Permitted Investments"), or in money market funds that are invested solely in Permitted Investments. Any net profit resulting from, or interest or income produced by, such investments will be payable to Parent, and will be refunded to Parent at Parent's option, provided that the Surviving Corporation will continue to be liable for any payments required to be made thereafter pursuant to Section 1.6(a) and this Section 1.9. None of Parent, Operating Sub, Sub, the Company, the Surviving Componation, or the Disbursing Agent will be liable t

(b) As soon as practicable after the Effective Time, but in any event within three business days after delivery of the Company's share register listing stockholders of record as of immediately before the Effective Time from the Company's transfer agent to the Disbursing Agent, the Disbursing Agent shall mail to each holder of record (other than Parent, Operating Sub, Sub, or any other direct or indirect subsidiary of Parent or the Company) of a certificate (a "Certificate") that immediately before the Effective Time represented issued and outstanding shares of Company Common Stock (other than holders of Dissenting Shares), a letter of transmittal (the "Letter of Transmittal") for return to the Disbursing Agent, and instructions for use in effecting the surrender of Certificates and to receive cash for each of such holder's shares of Company Common Stock under Section 1.6(a). The Letter of Transmittal will specify that delivery will be effected, and risk of loss will pass, only upon proper delivery of the Certificate to the Disbursing Agent. The Disbursing Agent, as soon as practicable following receipt of any such Certificate together with the Letter of Transmittal, duly executed, and any other items specified by the Letter of Transmittal, shall pay, by wire transfer, check, or draft, to the persons entitled thereto, the sum of the amounts determined by multiplying (i) the

number of shares of Company Common Stock formerly represented by the Certificate so surrendered by (ii) the Merger Consideration. All of the foregoing payments are subject to any required withholding of taxes by the Surviving Corporation and to the extent that amounts are so deducted and withheld, such deducted and withheld amounts will be treated for all purposes of this Agreement as having been paid to the person in respect of whom such withholding and deduction was made. No interest will be paid or accrued on the cash payable upon the surrender of the Certificates. If payment is to be made to a person other than the person in whose name the Certificate surrendered is registered, it will be a condition of payment that the Certificate so surrendered be properly endorsed or otherwise in proper form for transfer and that the person requesting the payment pays any transfer or other taxes required by reason of the payment to a person other than the registered holder of the Certificate surrendered or establish to the satisfaction of the Surviving Corporation that the tax has been paid or is not applicable.

(c) If any Certificate has been lost, stolen, or destroyed, upon the making of an affidavit of that fact by the person claiming the Certificate to have been lost, stolen, or destroyed, the amount to which such person would have been entitled under Section 1.9(b) but for failure to deliver the Certificate to the Disbursing Agent shall nevertheless be paid to such person, provided that the Surviving Corporation or the Disbursing Agent may, in its sole discretion and as a condition precedent to such payment, require such person to give Parent, the Surviving Corporation, and the Disbursing Agent a written indemnity agreement in form and substance reasonably satisfactory to the Surviving Corporation and the Disbursing Agent and, if deemed advisable in the sole discretion of the Surviving Corporation or the Disbursing Agent, a bond in such amount as the Surviving Corporation or the Disbursing Agent may direct as indemnity against any claim that may be brought against Parent, the Surviving Corporation, or the Disbursing Agent with respect to the Certificate alleged to have been lost, stolen, or destroyed.

1.10 No Further Rights or Transfers. At and after the Effective Time, all shares of Company Common Stock issued and outstanding immediately before the Effective Time will be canceled and cease to exist, and each holder of a Certificate that represented shares of Company Common Stock issued and outstanding immediately before the Effective Time will cease to have any rights as a stockholder of the Company with respect to the shares of Company Common Stock represented by the Certificate, except for the right to surrender the holder's Certificate in exchange for the payment provided under Section 1.6(a) or to perfect the holder's right to receive payment for such holder's shares under Section 262 of the Delaware Law and Section 1.7 if the holder has properly demanded and perfected and not withdrawn or lost the holder's right to receive payment for such holder's shares under Section 262 of the Delaware Law, and no transfer of shares of Company Common Stock issued and outstanding immediately before the Effective Time may be made on the stock transfer books of the Surviving Corporation.

Article II Closing

2.1 <u>Time and Place</u>. Subject to the provisions of Article VII, the closing (the "<u>Closing</u>") of the transactions contemplated hereby shall take place at the offices of Gowling Lafleur Henderson LLP, Toronto, Ontario, Canada, on the same business day as, and promptly following, completion of the meeting of stockholders referred to in Section 5.2(a) or at such other place or at such other date or time as Parent and the Company may mutually agree. The date on which the Closing occurs is the "<u>Closing Date</u>."

2.2 Deliveries at the Closing. At the Closing:

- (a) there shall be delivered to Parent, Operating Sub, Sub, and the Company the certificates and other documents and instruments the delivery of which is contemplated under Article VII;
- (b) Sub and the Company shall cause the Certificate of Merger to be filed as provided in Section 1.2 and shall take all other lawful actions and do any and all other lawful things necessary to cause the Merger to become effective; and
- (c) Parent shall provide evidence reasonably satisfactory to the Company that the amount designated as the Fund in Section 1.9 has been irrevocably deposited with the Disbursing Agent, subject to the rights of Parent to receive a refund of amounts remaining in the Fund six months after the Effective Time under Section 1.9.

Article III Representations and Warranties of the Company

The Company represents and warrants to Parent, Operating Sub and Sub, except as set forth in the disclosure schedule delivered by the Company to Parent, Operating Sub, and Sub contemporaneously herewith (the "Disclosure Schedule"), as follows:

3.1 Corporate Organization and Qualification. The Company is a corporation duly organized, validly existing, and in good standing under the laws of the State of Delaware and has the requisite corporate power and authority to own, lease, and operate all of its properties and assets and to carry on its business as it is now being conducted. Each of the subsidiaries of the Company is a corporation duly organized, validly existing, and in good standing under the laws of the jurisdiction of its incorporation and has the requisite corporate power and authority to own, lease, and operate all of its properties and assets and to carry on its business as it is now being conducted. Each subsidiary of the Company is incorporated in the jurisdiction set forth opposite its name in Section 3.1 of the Disclosure Schedule. Each of the Company and its subsidiaries is duly qualified or licensed as a foreign corporation to do business, and is in good standing, in each jurisdiction where the character of its properties owned, operated, or leased, or the nature of its activities, makes such qualification or licensing necessary, except such jurisdictions where failure to be so qualified or licensed has not had and

would not reasonably be expected to have a Material Adverse Effect (as defined in Section 7.1(d)). The Company has previously delivered to Parent a copy of its Certificate of Incorporation and Bylaws and copies of all similar organizational documents of its subsidiaries, in each case as amended to date and in effect as of the date of this Agreement. True and complete copies of all minute books of the Company and each of its subsidiaries, containing minutes of meetings held and actions taken by their respective boards of directors or any committees thereof during the period from their respective dates of incorporation to the date hereof, have been made available by the Company to Parent. All material actions of the boards of directors of the Company and each of its subsidiaries are reflected in such books.

3.2 Capitalization.

- (a) The authorized capital stock of the Company consists of 30 million shares of Company Common Stock and 10 million shares of preferred stock, par value US\$.0001 per share ("Company Preferred Stock," which, together with the Company Common Stock is the "Company Stock"). The rights, privileges, and preferences of the Company Common Stock are as stated in the Company's Certificate of Incorporation. On the date hereof, 11,642,351 shares of Company Common Stock were issued and outstanding, and no shares of Company Common Stock were held in the treasury of the Company. None of the shares of Company Preferred Stock has been designated as to class or series, and none of the shares of Company Preferred Stock are issued and outstanding or held in treasury by the Company. Except as set forth above in this Section 3.2(a), the Company has no other issued or outstanding shares of capital stock.
- (b) As of the date hereof, there are no outstanding subscriptions, options, warrants, or other rights to purchase Company Stock or any other capital stock or other equity securities of the Company or its subsidiaries are bound in respect of the Company Stock or other capital stock or other equity securities of the Company or its subsidiaries, whether issued or unissued, and no outstanding securities are convertible into or exchangeable for Company Stock or any other capital stock or other equity securities of the Company or its subsidiaries, except for Options to purchase up to an aggregate of 564,450 shares of Company Common Stock. There are no outstanding stock appreciation rights, phantom stock rights, performance shares, or other similar rights based upon the value of the Company Stock or any other capital stock or other equity securities of the Company or its subsidiaries. There are no declared but unpaid or accrued dividends on the Company Common Stock.
- (c) Section 3.2(c) of the Disclosure Schedule lists for each of the Options, as of the date hereof, (i) the person holding the Option; (ii) the number of shares of Company Common Stock subject to the Option; (iii) the per-share exercise price of the Option; (iv) the Company option plan under which the Option was granted; and (v) the Grant Date (as defined in Section 3.2(g)) of the Option.
 - (d) All of the outstanding shares of capital stock or other equity securities of the Company and its subsidiaries are validly issued, fully paid, and nonassessable.

- (e) Section 3.2(e) of the Disclosure Schedule sets forth the name of each subsidiary of the Company and the percentage of the outstanding capital stock or other equity securities of such subsidiary owned, directly or indirectly, by the Company free and clear of all liens, mortgages, security interests, encumbrances, conditions, pledges, charges, or restrictions of any nature ("Encumbrances") other than restrictions on transfer imposed by applicable securities laws. The Company owns no other equity securities of, or equity interest in, any other entity.
- (f) None of the outstanding shares of capital stock or other equity securities of the Company or any of its subsidiaries or the Options were granted in violation of preemptive or similar rights. Other than the Voting Agreement and the Irrevocable Proxies, there are no voting trusts, proxies, voting agreements, rights plans, anti-takeover plans, registration rights agreements, or similar agreements or understandings to which the Company or any of its subsidiaries is a party or of which the Company otherwise has knowledge with respect to the voting of capital stock of the Company. There are no outstanding contractual obligations of the Company or any of its subsidiaries (i) restricting the transfer of; (ii) affecting the voting rights of; (iii) requiring the repurchase, redemption, or disposition of, or containing any right of first refusal with respect to; (iv) requiring the registration for sale of; or (v) granting any preemptive or antidilutive right with respect to, any shares of Company Common Stock or any capital stock of, or other equity interests in, the Company or any of its subsidiaries.
- (g) With respect to the Options, (i) each Option intended to qualify as an "incentive stock option" under Section 422 of the Internal Revenue Code of 1986, and the rules and regulations adopted pursuant thereto (the "Code"), so qualifies; (ii) each grant of an Option was duly authorized no later than the date on which the grant of such Option was by its terms to be effective (the "Grant Date") by all necessary corporate action, including, as applicable, approval by the Company board of directors (the "Company Board") (or a duly constituted and authorized committee thereof), or a duly authorized delegate thereof, and any required stockholder approval by the necessary number of votes or written consents; (iii) each such grant was made in accordance with the terms of the applicable Company stock option plan, the United States Securities Exchange Act of 1934 (the "Exchange Act") and all other Applicable Laws, including the rules of the Toronto Stock Exchange; (iv) the per share exercise price of each Option was not less than the fair market value of a share of Company Common Stock on the applicable Grant Date; and (v) each such grant was properly accounted for in all material respects in accordance with United States generally accepted accounting principles ("GAAP") in the financial statements (including the related notes) of the Company and disclosed in the Company SEC Reports (as defined in Section 3.4) and all other Applicable Laws. The Company has not granted, and there is no and has been no Company policy or practice to grant, Options prior to, or otherwise coordinate the grant of Options with, the release or other public announcement of material information regarding the Company or any of its subsidiaries or their financial results or prospects. The stock option plans under which the Options were granted permit, or as of the Closing

Date will permit, the treatment of such Options in the Merger as contemplated by Section 1.8.

- (h) For purposes of this Agreement:
- (i) "Applicable Laws" means, with respect to any person, any international, national, federal, state, provincial, or local law (statutory, common, or otherwise), constitution, treaty, convention, ordinance, code, rule, regulation, or other similar requirement enacted, adopted, promulgated, or applied by a Governmental Authority that is binding upon or applicable to such person, as amended unless expressly specified otherwise.
- (ii) "Governmental Authority" means (A) any government or any state, province, department, local authority, or other political subdivision thereof; (B) any governmental body, agency, authority (including any taxing authority or transgovernmental or supranational entity or authority), minister, or instrumentality (including any court, arbitral body, or other tribunal) exercising executive, legislative, judicial, regulatory, or administrative functions of or pertaining to government; or (C) any stock exchange (including the Toronto Stock Exchange).

3.3 Authority; Non-Contravention.

- (a) The Company has the corporate power and authority to execute and deliver this Agreement and, subject to obtaining the adoption and approval of this Agreement by the stockholders of the Company, to consummate the transactions contemplated hereby. The execution, delivery, and performance of this Agreement by the Company have been duly and effectively authorized by the Company Board. As of the date hereof, the Company Board has determined to recommend unanimously the adoption and approval of this Agreement and the Merger to the stockholders of the Company. Except for the adoption and approval of this Agreement and the Merger by the stockholders of the Company as provided in Section 5.2(a), no further corporate action is necessary on the part of the Company to authorize the consummation of the transactions contemplated hereby. This Agreement has been duly and validly executed and delivered by the Company and, assuming the accuracy of the representations and warranties of Parent, Operating Sub, and Sub set forth in the first two sentences of Section 4.2(a), constitutes the valid and binding agreement of the Company, enforceable against the Company in accordance with its terms, except to the extent that enforceability may be limited by applicable bankruptcy, insolvency, or similar laws affecting the enforcement of creditors' rights generally and by general principles of equity (regardless of whether enforcement is sought in a court of law or equity) (the "Enforcement Limitations").
- (b) Neither the execution and delivery of this Agreement by the Company, nor the consummation by the Company or any of its subsidiaries of the transactions contemplated hereby, will (i) conflict with or result in a breach of the corporate charter or

bylaws or similar organizational documents, as currently in effect, of the Company or any of its subsidiaries; (ii) require the consent or approval of, or any filing with, any Governmental Authority having jurisdiction over any of the businesses or material assets of the Company or any of its subsidiaries or violate any order, writ, injunction, decree, statute, rule, or regulation applicable to the Company or any of its subsidiaries or any of their properties or assets; or (iii) result in a breach of, or constitute a default or an event that, with the passage of time or the giving of notice or both, would constitute a default, give rise to a right of termination, cancellation or acceleration, create any entitlement to any payment or benefit, require notice to or the consent of any third party, or result in the creation of a lien on any of the properties or assets of the Company or any of its subsidiaries under, any other instrument, contract, or agreement to which the Company or any of its subsidiaries is a party or by which any of them or any of their properties or assets may be bound, except, (I) in the case of clause (ii), (A) such reports under Section 15(d) of the Exchange Act as may be required in connection with this Agreement and the transactions contemplated hereby and (B) the filing of the Certificate of Merger with the Secretary of State of the State of Delaware and appropriate documents with the relevant authorities of other states in which the Company is qualified to do business as a foreign corporation, and (II) in the case of clause (iii), such breach or default would not reasonably be expected to have a Material Adverse Effect.

(c) No antitrust or competition-law notices, filings, or approvals by or with respect to the Company or any of its subsidiaries, including those under the United States Hart-Scott-Rodino Antitrust Improvements Act of 1976, the Investment Canada Act, or the Competition Act (Canada), are required in connection with the Merger.

3.4 Securities Reports.

(a) The Company has heretofore made available to Parent (i) a draft dated June 16, 2008 of the Company's Annual Report on Form 10-K for the fiscal year ended March 31, 2008 (the "2008 Form 10-K"), and, (ii) in the form filed with the United States Securities and Exchange Commission (the "SEC"), all reports or registration statements and all other filings made by the Company with the SEC since June 30, 2006 (collectively with the 2008 Form 10-K, the "SEC Reports"). No SEC Report (including any document incorporated by reference therein), as of its filing date or, if amended, as of the date of the last such amendment, or, with respect to the 2008 Form 10-K, as of June 16, 2008 (other than in connection with any required subsequent disclosure relating to the transaction contemplated hereby), contained or will contain any untrue statement of a material fact or omitted or will omit to state any fact required to be stated therein or necessary in order to make the statements made therein, in the light of the circumstances under which they were made, not misleading, and each SEC Report at the time of its filing complied or will comply as to form in all material respects with Applicable Laws. Since November 2, 2006, the Company has filed in a timely manner all reports that it was required to file with the SEC under the Exchange Act and the rules and regulations of the SEC.

- (b) Each of the consolidated financial statements contained in the SEC Reports (i) comply as to form, as of their respective filing dates with the SEC, in all material respects with the applicable accounting requirements and the published rules and regulations of the SEC with respect thereto; (ii) was prepared in accordance with GAAP applied on a consistent basis throughout the periods involved (except as may be indicated in the notes thereto); and (iii) fairly presented in all material respects the consolidated financial position of the Company and its subsidiaries as at the respective dates thereof and the consolidated results of operations and cash flows and changes in stockholders' equity of the Company and its subsidiaries for the periods indicated, subject in the case of quarterly financial statements to normal year-end adjustments and except that the quarterly financial statements do not contain all of the footnote disclosures required by GAAP to the extent permitted by the rules and regulations of the SEC.
- (c) The Company's system of internal controls over financial reporting is reasonably sufficient in all material respects to provide reasonable assurance (i) that transactions are recorded as necessary to permit preparation of financial statements in conformity with GAAP; (ii) that receipts and expenditures are executed in accordance with the authorization of management; and (iii) regarding prevention or timely detection of the unauthorized acquisition, use, or disposition of the Company's assets that would materially affect the Company's financial statements. No significant deficiency or material weakness was identified in management's assessment of internal controls as of March 31, 2008 (nor has any such deficiency or weakness since been identified).
- (d) The Company's "disclosure controls and procedures" (as defined in Rules 13a-15(e) and 15d-15(e) of the Exchange Act) are reasonably designed to ensure that (i) all information (both financial and non-financial) required to be disclosed by the Company in the reports that it files or submits under the Exchange Act is recorded, processed, summarized and reported to the individuals responsible for preparing such reports within the time periods specified in the rules and forms of the SEC; and (ii) all such information is accumulated and communicated to the Company's management as appropriate to allow timely decisions regarding required disclosure and to make the certifications of the principal executive officer and principal financial officer of the Company required under the Exchange Act with respect to such reports.
 - (e) The Company is in material compliance with the applicable listing and corporate governance rules and regulations of the Toronto Stock Exchange.
- (f) Except as permitted by the Exchange Act, since November 2, 2006, neither the Company nor any of its affiliates has made, arranged, or modified (in any material way) any extensions of credit in the form of a personal loan to any executive officer or director of the Company.
- 3.5 <u>Undisclosed Liabilities</u>. There are no material liabilities, either individually or in the aggregate, of the Company or any of its subsidiaries of any kind whatsoever, whether accrued, contingent, absolute, determined, determinable or otherwise, other than (a) liabilities disclosed or set forth in the Company's balance sheet included in the 2008 Form 10-K (the

"Balance Sheet"); (b) liabilities incurred in the ordinary course of business since the date of the Balance Sheet; (c) liabilities under, or required to be incurred under, this Agreement; and (d) liabilities (other than those in default) (i) under contracts and agreements set forth in Section 3.9 of the Disclosure Schedule or the non-disclosure of which therein does not constitute a misrepresentation under Section 3.9 of this Agreement or (ii) under Employee Plans set forth in Section 3.14 of the Disclosure Schedule.

3.6 Absence of Changes. Except for liabilities incurred in connection with this Agreement and the transactions contemplated hereby, and except as disclosed in the SEC Reports filed before the date of this Agreement, from the date of the Balance Sheet through the date of this Agreement, the Company and its subsidiaries have conducted their business only in the ordinary course consistent with past practice, and there has not been (a) any damage, destruction or loss, whether covered by insurance or not, that has had or would reasonably be expected to have a Material Adverse Effect; (b) any other event, development or condition (financial or otherwise) of any character or any operations or results of operations that has had or would reasonably be expected to have a Material Adverse Effect; or (c) any other action or event that would have required the written consent of Parent under Section 5.1 had such action or event occurred after the date of this Agreement.

3.7 Taxes.

- (a) For purposes of this Agreement, the following definitions shall apply:
- (i) "Tax" or "Taxes" means all tax imposed by any Governmental Authority, including national, provincial, state, local, or foreign taxes and other taxes on income, estimated income, alternative or add-on minimum, gross receipts, profits, business, license, occupation, stamp, premium, value added, consumption, utility, franchise, service, personal and real property (including special assessments or charges), sales, use, transfer, gains, excise, severance, environmental, unclaimed property, employment, unemployment, payroll, withholding, disability, social security, minimum tax, capital stock, registration, or any other tax, custom duty, ad valorem levy, governmental fee, or other like assessment or charge of any kind, together with any interest or any penalty, addition to tax, or additional amount, whether disputed or not, and including any liability for the taxes of any person as a transferee, successor, or agent, by contract, or otherwise.
- (ii) "Tax Returns" means all returns, forms, computations, declarations, elections, and claims for refund relating to Taxes, including any schedules or attachments thereto and any amendments thereof.
- (b) (i) All Tax Returns required to be filed by, or with respect to, the Company and its subsidiaries have been duly and timely filed (taking into account any extensions); (ii) the information included in the Tax Returns filed by or with respect to the Company and its subsidiaries is complete and accurate in all material respects; (iii) all Taxes due and payable by the Company and its subsidiaries have been timely paid; (iv)

no Taxes are due by the Company and its subsidiaries in any jurisdiction in which Tax Returns have not been filed; (v) no action, suit, examination, proceeding, investigation, audit, or claim is now proposed, pending, or, to the knowledge of the Company, threatened with respect to the Company or any of its subsidiaries in respect of any Tax; (vi) no notice or claim has been made by a Governmental Authority in a jurisdiction where the Company or any of its subsidiaries does not file a particular Tax Return that it is or may be subject to that particular Tax in that jurisdiction; (vii) there are no liens for Taxes upon the assets of the Company or any of its subsidiaries, other than liens for Taxes not yet due or payable; and (viii) the Balance Sheet reflects an adequate reserve for all Taxes for which the Company or any of its subsidiaries may be liable for all taxable periods and portions thereof through the date thereof.

- (c) The Company and its subsidiaries have withheld and paid all Taxes required to have been withheld under Applicable Law.
- (d) Neither the Company nor its subsidiaries has any liability for Taxes of any other person (other than the Company or any subsidiary) as a transferee, successor, or agent, by contract, or otherwise. Neither the Company nor any of its subsidiaries has ever been a party to or bound by any Tax-indemnity, Tax-allocation, or Tax-sharing agreement.
- (e) Neither the Company nor any of its subsidiaries has agreed or is required to include any material item of income in, or exclude any material item of deduction from, taxable income for any taxable period as a result of any (i) change in method of accounting for a taxable period; (ii) agreement with any Governmental Authority with regard to the Tax liability of the Company or any of its subsidiaries; (iii) installment sale or open transaction disposition by the Company or the relevant subsidiary; or (iv) prepaid amount.
- (f) The Company and each of its subsidiaries is duly and properly registered in each country and territory where required by Applicable Law for VAT and any other equivalent local or regional Taxes for which registration is required by Applicable Law.
- (g) There is no agreement, plan, arrangement, or other contract covering any current or former employee or independent contractor of the Company or any of its subsidiaries that would reasonably be expected to give rise to the payment of any amount that would not be deductible pursuant to any applicable Tax law or regulation.

3.8 Properties.

(a) The Company and its subsidiaries have good and marketable title to all property, assets, and rights reflected in the Balance Sheet or acquired by the Company and its subsidiaries after the date of the Balance Sheet (except for inventory and obsolete equipment sold or otherwise disposed of in the ordinary course of business) or otherwise purported to be owned by them, and have a valid leasehold interest in or other right to use

all other property, assets, and rights used in their businesses, free and clear of all Encumbrances of any kind or character other than:

- (i) liens for Taxes not yet due and payable;
- (ii) mechanic's, warehousemen's, materialmen's, landlord's, or similar liens securing obligations incurred in the ordinary course of business that are not yet due and payable;
- (iii) Encumbrances on real property in the nature of zoning restrictions, easements, rights of way, encroachments, restrictive covenants, and other similar rights or restrictions that were not incurred in connection with the borrowing of money or the obtaining of advances or credit and that do not, individually or in the aggregate, materially detract from the value of the properties subject thereto or affected thereby or materially impair present business operations at such properties; and
 - (iv) existing Encumbrances disclosed in the Balance Sheet (or in the notes thereto).
- (b) Section 3.8(b) of the Disclosure Schedule sets forth a complete list of all real property and interests in real property owned in fee by the Company or any of its subsidiaries (individually, an "Owned Real Property" and collectively, the "Owned Real Properties"). With respect to each parcel of Owned Real Property:
 - (i) Except for noncompliance that would not reasonably be expected to have a Material Adverse Effect, the use and occupancy of the Owned Real Properties are in compliance with all Applicable Laws and all applicable insurance requirements.
 - (ii) There are no leases, subleases or occupancy agreements in effect with respect to the Owned Real Properties. There are no pending or, to the knowledge of the Company, threatened or contemplated actions or proceedings regarding condemnation or other eminent domain actions or proceedings affecting the Owned Real Properties or any part thereof or of any sale or other disposition of the Owned Real Properties or any part thereof in lieu of condemnation.
- (c) Section 3.8(c) of the Disclosure Schedule sets forth a complete list of all real property and interests in real property leased or occupied by the Company or any of its subsidiaries, or which the Company or any of its subsidiaries has the right to occupy, now or in the future (each, a "Real Property Lease," and the real properties specified in such leases being referred to as the "Leased Real Property"). With respect to each parcel of Leased Real Property:
 - (i) Each Real Property Lease is legal, valid, and binding and in full force and effect (subject to the Enforcement Limitations), and neither the

Company nor any of its subsidiaries nor, to the knowledge of the Company, any other party thereto is in material default under or in respect of any Real Property Lease.

- (ii) None of the Company, any of its subsidiaries, nor any other party to any Real Property Lease is in breach or default under such Real Property Lease, except for (A) such defaults and events as to which requisite waivers or consents have been obtained, and (B) breaches or defaults that would not reasonably be expected to have a Material Adverse Effect.
- (iii) No Real Property Lease requires the consent of any landlord as a result of the transactions contemplated by this Agreement, except where the failure to obtain such consent would not reasonably be expected to have a Material Adverse Effect.
 - (iv) The Company has made available to Parent correct and complete copies of each Real Property Lease.
- (d) All of the land, buildings, structures, and other improvements used by the Company or any of its subsidiaries in the conduct of their respective businesses are included in the Owned Real Property and the Leased Real Property.
- (e) Neither the Company nor any of its subsidiaries is a party to any development, incentive, or other agreement with any Governmental Authority that limits in any material respect the right of the Company or any of its subsidiaries to protest property-related Taxes, establishes minimum property-related Taxes, or requires continued business operation at any particular location.
- 3.9 Contracts. All Contracts in effect on the date hereof have been either included as an exhibit to an SEC Report filed before the date of this Agreement or made available to Parent and listed in Section 3.9 of the Disclosure Schedule. All Contracts are legal, valid, and binding and in full force and effect (subject to the Enforcement Limitations), and neither the Company nor any of its subsidiaries nor, to the knowledge of the Company, any other party thereto is in material default under or in respect of any Contract. As used herein, "Contract" means (a) each "material contract" (as such term is defined in Item 601(b)(10) of Regulation S-K promulgated by the SEC); and (b) each of the following other agreements or contracts, whether written or oral, to which the Company or any of its subsidiaries is a party or by which any of them or their properties or assets are bound:
 - (i) each effective employment agreement, severance agreement, non-competition agreement, or indemnification agreement with any director, employee, or former employee of the Company or any of its subsidiaries;
 - (ii) each contract, whether as licensor or licensee, for the license of any patent, know-how, trademark, trade name, service mark, copyright, software, or other

intangible asset (other than non-negotiated licenses of generally available commercial software);

- (iii) each loan agreement, indenture or other instrument, contract, or agreement under which any money has been borrowed or loaned or under which any note, bond, or other evidence of indebtedness has been issued and remains outstanding, each guaranty, indemnification, or assumption agreement, and each contract to reimburse any maker of a letter of credit or banker's acceptance;
- (iv) each mortgage, contract for deed, security agreement, conditional sales contract, financing, "synthetic" or capitalized lease, or similar agreement that effectively creates an Encumbrance on any assets of Company or any of its subsidiaries (other than any purchase money security interest, conditional sales contract, capitalized lease, or similar agreement that creates an Encumbrance only on tangible personal property and the unpaid obligations of the Company or any of its subsidiaries under which are C\$250,000 or less);
- (v) each contract restricting the Company or any of its subsidiaries from engaging in business or from competing with any other persons, or pursuant to which any benefit or right is required to be given or lost as a result of so engaging or competing;
 - (vi) each partnership or joint venture agreement;
- (vii) each agreement for the purchase or sale of products or services (other than purchase or sales orders entered into in the ordinary course of business on an order-by-order basis) under which the undelivered balance of such products or services has a price in excess of C\$250,000;
 - (viii) each agreement for capital expenditures the unpaid obligations of the Company or any of its subsidiaries under which exceed C\$250,000;
 - (ix) each lease of tangible personal property the unpaid obligations of the Company or any subsidiary under which exceed C\$250,000;
- (x) each agreement for the purchase or sale of any business, division, or subsidiary by or to the Company or any of its subsidiaries not yet consummated or under which the Company or any of its subsidiaries has any continuing indemnification obligations;
- (xi) each agreement with any officer or director of the Company or any of its subsidiaries, any beneficial owner of five percent or more of the outstanding Company Common Stock, any ascendant, descendent, sibling or spouse of any such officer, director or beneficial owner, or any trust, partnership, corporation or other entity in which any of such persons has at least a five percent equity interest (collectively, "Associates");

- (xii) each option, right of first refusal, or right of first offer with respect to the sale, purchase, or leasing of any real property;
- (xiii) each agreement that contains any "take or pay" provisions that obligate the Company or any of its subsidiaries to make minimum periodic payments or payment commitments to the Company's or any of its subsidiaries' carrier service providers for communications bandwidth;
- (xiv) each agreement that (A) grants any exclusive license or supply or distribution agreement or other exclusive rights; (B) grants any "most favored nation" rights, rights of first refusal, rights of first negotiation, or similar rights with respect to any product or service; or (C) contains any provisions that requires the purchase of all or a given portion of the Company's or any of its subsidiaries' requirements from a given third party, or any similar provision;
 - (xv) each agreement that relates to any derivatives or hedging transaction;
- (xvi) each agreement that gives rise to a right of termination, cancellation, or acceleration, creates any entitlement to any material payment or benefit, requires notice to or the consent of any third party, or results in the creation of an Encumbrance on any of the properties or assets of the Company as a result of the transactions contemplated by this Agreement; and
 - (xvii) each agreement entered into other than in the ordinary course of business.

3.10 Intellectual Property.

- (a) Other than Intellectual Property (as defined below) that is licensed by the Company or any of its subsidiaries ("Licensed Intellectual Property"), Section 3.10(a) of the Disclosure Schedule sets forth a true and correct list of all: (i) trademarks, service marks, trade names, logos, internet domain names, and other symbols, names or marks used to identify products or services; (ii) patents and other industrial property rights; (iii) works of authorship, including copyrights and moral rights; (iv) computer programs and software, including source code, object code, firmware, and HTML code (collectively "Software"); (v) databases, compilations, customer lists, data, trade secrets, confidential information, and know-how; (vi) technology, inventions, processes, formulae, algorithms, models, and methodologies; (vii) all rights in or relating to the foregoing; and (viii) any other intellectual property rights, rights of publicity or privacy (collectively "Intellectual Property") that are held or used by the Company or any of its subsidiaries (collectively, the "Owned Intellectual Property") that are: (A) the subject of a registration or an application for a registration with a Governmental Authority; or (B) material to the operation of the business of the Company or any of its subsidiaries.
 - (b) To the knowledge of the Company, no Intellectual Property is used or held for use by the Company or any of its subsidiaries in the conduct of its business, except

the Licensed Intellectual Property that is identified in Section 3.10(b) of the Disclosure Schedule and any Owned Intellectual Property.

- (c) Neither the Company nor any of its subsidiaries has (i) defaulted under any license of Licensed Intellectual Property; (ii) knowledge of circumstances that would reasonably be expected to give rise to any such default; or (iii) knowledge of circumstances that are causing or would be reasonably expected to cause the loss or impairment of any Owned Intellectual Property or Licensed Intellectual Property.
- (d) (i) The Company exclusively owns all right, title, and interest in and to the Owned Intellectual Property free and clear of any Encumbrance; and (ii) neither the Company nor any of its subsidiaries has granted any person any right in or to the Owned Intellectual Property or the Licensed Intellectual Property other than in the ordinary course of the business of the Company or its subsidiaries in a manner that would not reasonably be expected to have a Material Adverse Effect.
- (e) Each of the Company and its subsidiaries owns or otherwise has all Intellectual Property rights necessary for the conduct of its business as currently conducted and as currently proposed to be conducted by the Company and its subsidiaries. Neither the execution, delivery, or performance of this Agreement (or any of the ancillary agreements) nor the consummation of any of the transactions contemplated by this Agreement (or any of the ancillary agreements) will, with or without notice or lapse of time, result in, or give any other person the right or option to cause or declare, (i) a loss of, or Encumbrance on, any material Owned Intellectual Property or Licensed Intellectual Property; (iii) a material breach of or default under any license of Licensed Intellectual Property; (iii) the release, disclosure, or delivery of any Owned Intellectual Property by or to any escrow agent or other person; or (iv) the grant, assignment, or transfer to any other person of any license or other right or interest under, to, or in any of the Owned Intellectual Property.
- (f) Neither the Company nor any of its subsidiaries: (i) infringes, has infringed, misappropriated, or otherwise violated any Intellectual Property of any other person; (ii) has made use of any Intellectual Property in violation of the rights of any other person; or (iii) has knowledge of any allegations by any other person to the contrary or any facts or circumstances that could reasonably give rise to such an allegation.
- (g) Neither the Company nor any of its subsidiaries has knowledge of any pending Intellectual Property of any other person that, if issued or registered, would be infringed upon by the current operations of the Company or any of its subsidiaries.
- (h) To the knowledge of the Company and each of its subsidiaries, no person is infringing, misappropriating, or otherwise violating in any material respect the Owned Intellectual Property or the Licensed Intellectual Property or is making use of any Intellectual Property in violation of the rights of any of the Company or any of its subsidiaries. The Company and each of its subsidiaries have taken all steps reasonably required to maintain applications and registrations for the Owned Intellectual Property in

full force and effect, including timely payment of all fees and timely filing of all documents required under Applicable Laws. None of the Company or any of its subsidiaries or, to the knowledge of the Company or any of its subsidiaries, any licensor or licensee thereof has used or enforced, or failed to use or enforce, any of the Owned Intellectual Property or the Licensed Intellectual Property in any manner that is reasonably likely to limit its validity or result in its invalidity, or has received any notice that any of the Owned Intellectual Property or the Licensed Intellectual Property has been declared unenforceable or otherwise invalid by any Governmental Authority.

- (i) None of the Software used, marketed, distributed, provided, or licensed by the Company or any of its subsidiaries (collectively, "Company Software") (i) contains any bug, defect, or error that affects the use, functionality, or performance of such Company Software or any product or system containing or used in conjunction with such Company Software in each case in a way that would reasonably be expected to have a Material Adverse Effect; or (ii) fails to comply with any applicable warranty or other contractual commitment relating to the use, functionality, or performance of such Company Software, in each case in a way that would be reasonably likely to have a Material Adverse Effect. No Company Software contains any "back door," "drop dead device," "time bomb," "Trojan horse," "virus," or "worm" (as such terms are commonly understood in the software industry) or any other code designed or intended to have, or capable of performing, any of the following functions: (A) disrupting, disabling, harming, or otherwise impeding in any manner the operation of, or providing unauthorized access to, a computer system or network or other device on which such code is stored or installed; or (B) damaging or destroying any data or file without the user's consent, in each case in a way that would reasonably be expected to have a Material Adverse Effect.
- (j) No product or service used, sold, distributed, licensed, leased, provided, or offered by the Company or any of its subsidiaries ("Company Product") contains, is derived from, is distributed with, or is being or was developed using Software that is licensed under any terms that (i) impose or could impose a requirement or condition that any Company Product or part thereof (A) be disclosed or distributed in source code form, (B) be licensed for the purpose of making modifications or derivative works, or (C) be redistributable at no charge; or (ii) otherwise impose or could impose any other material limitation, restriction, or condition on the right or ability of the Company or any of its subsidiaries to use, sell, distribute, license, lease, provide, or offer any Company Product.
- (k) Section 3.10(k) of the Disclosure Schedule identifies each privacy policy and data security policy of the Company or any of its subsidiaries (collectively, "Company Privacy Policy") in effect at any time and identifies, with respect to each Company Privacy Policy, (i) the period of time during which such privacy policy was or has been in effect; (ii) whether the terms of a later Company Privacy Policy apply to the data or information collected under such privacy policy; and (iii) if applicable, the mechanism (such as opt-in, opt-out, or notice only) used to apply a later Company Privacy Policy to data or information previously collected under such privacy policy. Each of the Company and its subsidiaries have complied in all material respects at all

times with all of the Company Privacy Policies and with all Applicable Laws pertaining to privacy and data.

- (I) Section 3.10(I) of the Disclosure Schedule identifies and describes each distinct electronic or other database containing any information relating to an identified or identifiable natural person ("Personal Data") maintained by or for the Company or any of its subsidiaries at any time (the "Personal Databases"). No material breach or violation of any Company Privacy Policy relating to the Personal Databases has occurred, and there has been no unauthorized or illegal use of or access to any of the Personal Data in a way that would reasonably be expected to have a Material Adverse Effect.
- (m) The computer hardware, Software, network infrastructure, communications networks and facilities that the Company or any of its subsidiaries uses, and other system support mechanisms related to the foregoing (collectively, "IT Systems"): (i) are functioning properly and in accordance with all applicable specifications in all material respects; (ii) have in all material respects sufficient capacity, processing power and performance to meet the current business requirements and the currently expected business requirements of each of the Company and its subsidiaries without material degradation in quality, speed or functionality; and (iii) are in all material respects properly documented and covered by adequate maintenance, disaster recovery and business continuity agreements. Each of the Company and its subsidiaries has implemented procedures and safeguards for ensuring the security of its IT Systems and the data contained therein that are compliant with industry standards, including compliance with the Payment Card Industry Data Security Standard. Neither the Company nor any of its subsidiaries is in default of any representations, warranties, or covenants relating to the performance or the capabilities of the IT Systems or applications executed thereon made by it to any of its customers under any agreement in a way that would reasonably be expected to have a Material Adverse Effect.
- (n) The Company has made available to Parent a complete and accurate copy of each Intellectual Property related standard form of agreement ("Company IP Contract") used by the Company and any of its subsidiaries at any time, including each standard form of (i) employee agreement containing any assignment or license of Intellectual Property; (ii) consulting or independent contractor agreement containing any Intellectual Property assignment or license; (iii) confidentiality or nondisclosure agreement(s); and (iv) authorization to use Owned Intellectual Property, Licensed Intellectual Property, or any aspect of the IT Systems. Section 3.10(n) of the Disclosure Schedule accurately identifies each Company IP Contract that deviates in any material respect from the corresponding standard form agreement.
- (o) Each person who is or was an employee or contractor of the Company or any of its subsidiaries, or any of their respective contractors, and who is or was involved in the creation or development of any Intellectual Property for the Company or any of its subsidiaries, has signed a valid, enforceable agreement (in substantially the form made available to Parent) containing an assignment of Intellectual Property to the Company or

such subsidiary and confidentiality provisions protecting the Intellectual Property of the Company and its subsidiaries. No current or former stockholder, officer, director, or employee of the Company or any of its subsidiaries has any claim, right (whether or not currently exercisable), or interest to or in any Intellectual Property used by the Company or any of its subsidiaries.

- (p) Each of the Company and its subsidiaries have taken all reasonable steps to maintain the confidentiality of and otherwise protect and enforce their rights in all proprietary information pertaining to each of the Company and its subsidiaries.
- 3.11 <u>Litigation</u>. As of the date hereof, there are no claims, litigation, arbitrations, administrative proceedings, abatement orders, or investigations of any kind pending or, to the knowledge of the Company, threatened in writing against the Company or any of its subsidiaries, or, to the knowledge of the Company, against any of their respective officers, employees or directors in connection with the business or affairs of the Company or any of its subsidiaries. As of the date hereof, there are no judgments, orders, writs, injunctions, decrees, indictments, subpoenas, or civil investigative demands or awards against the Company or any of its subsidiaries.

3.12 Compliance with Laws.

- (a) Each of the Company and its subsidiaries has complied in all material respects with, and is not in material default under or in material violation of, any Applicable Laws or any other governmental restrictions, orders, judgments or decrees applicable to it or its properties or assets, and, as of the date hereof, to the Company's knowledge, there have been no allegations by any Governmental Authority or private persons to the contrary.
- (b) Neither the Company nor any of its subsidiaries, nor anyone acting on their behalf, has made any material payments or otherwise provided any material benefits, direct or indirect, to any customer, supplier, Governmental Authority, or other person, or any employee or agent thereof, for the purpose of acquiring purchase or sales relationships, licenses, franchises, permits or other governmental authorizations, or for any other purpose, that are unlawful in any material respect.
- (c) The Company and each of its subsidiaries has conducted its export transactions in accordance in all material respects with all applicable provisions of export control laws and regulations. Without limiting the foregoing, (i) the Company and each of its subsidiaries is in material compliance with the terms of all export licenses or other approvals applicable to the Company or such subsidiary; and (ii) there are no pending or, to the knowledge of the Company, threatened claims against the Company or any of its subsidiaries with respect to such export licenses or other approvals.
 - (d) Neither the Company nor any of its subsidiaries is a franchisee or franchisor, or is or may be deemed to be a franchisee or franchisor under any Applicable Law.

- 3.13 <u>Licenses and Permits</u>. To the knowledge of the Company, each of the Company and its subsidiaries has in full force and effect all licenses, franchises, permits and other governmental authorizations necessary to permit it to lawfully conduct its business in the manner presently conducted and to own and use its properties and assets in the manner presently owned and used, and neither the Company nor any of its subsidiaries is in violation of any such material license, franchise, permit or other governmental authorization. No such license, franchise, permit or other governmental authorization will terminate or lapse as a result of the consummation of the transactions contemplated by this Agreement.
 - **3.14** Employee Plans. Except as otherwise provided in the Disclosure Schedule:
 - (a) Each (i) employee pension benefit plan ("Pension Plan"), as such term is defined in Section 3 of the United States Employee Retirement Income Security Act of 1974, and the rules and regulations adopted pursuant thereto ("ERISA"); (ii) employee welfare benefit plan ("Welfare Plan"), as such term is defined in Section 3 of ERISA; (iii) group retirement savings plan as defined in Section 146(1) of the Income Tax Act (Canada) ("ITA"); (iv) deferred profit sharing plan as defined in Section 147(1) of the ITA; (v) registered pension plan, retirement compensation arrangement, employee profit sharing plan, salary deferral arrangement or employee benefit plan as defined in Section 248(1) of the ITA; or (vi) other deferred compensation, bonus, incentive, stock incentive, option, stock purchase, severance, pension, health, medical, dental, disability, life insurance, or other employee benefit plan, agreement, commitment or arrangement that is maintained by the Company or any of its Benefits Affiliates (as defined in Section 3.14(k)), or to which the Company or any of its Benefits Affiliates contributes or is under any obligation to contribute, or with respect to which the Company or any of its Benefits Affiliates has any liability (whether current or contingent) (each, an "Employee Plan" and collectively, the "Employee Plans") is listed in Section 3.14 of the Disclosure Schedule, and a copy of each Employee Plan that is a written plan has been made available to Parent. In addition, copies of the most recent determination or opinion letter issued by the Internal Revenue Service, registration letter issued by the Canada Revenue Agency or any pension authority in Canada and, if applicable, the most recent actuarial reports or valuations with respect to each Pension Plan or registered pension plan in Canada, copies of the most recent summary plan description for each Pension Plan and each Welfare Plan, copies of the employee booklets for any other Employee Plans, copies of any trust agreement, insurance contract or other funding or investment arrangements for the benefits under each Employee Plan, and copies of the annual reports (e.g., Form 5500 Series in the US, the Pension Plan Information and Tax Return filed in Canada or the annual information return filed with the pension authority of the Province or the Office of Superintendent of Financial Institutions in Canada, as applicable) required to be filed with any Governmental Authority for each Employee Plan for the three most recent plan years of each such plan, have been made available to Parent.
 - (b) Each of the Company and its Benefits Affiliates has made on a timely basis all contributions or payments required to be made by it in all material respects

under the terms of the Employee Plans, ERISA, Code, ITA, applicable pension legislation in Canada or other Applicable Laws.

- (c) Each Employee Plan (and any related trust or other funding instrument) has been administered in all respects in compliance with its terms and in both form and operation is in compliance in all material respects with the applicable provisions of ERISA, Code, ITA, the pension standards legislation of any applicable Province or Canada and other Applicable Laws, and all reports required to be filed with any Governmental Authority with respect to each Employee Plan have been timely filed. The Company has no knowledge of facts that would be reasonably expected to cause the Internal Revenue Service to disqualify any Pension Plan that is intended to be a tax-qualified plan under Section 401(a) of the Code. Each Employee Plan in Canada that is subject to registration under the ITA or the pension standards legislation of any Province or Canada has been so registered, and nothing has occurred since the date of such registration that would be reasonably expected to cause the Canada Revenue Agency or the pension authority to revoke the registration of any such Employee Plan.
- (d) There are no claims (other than routine claims for benefits) or proceedings pending or, to the knowledge of the Company, threatened by the Internal Revenue Service, the U.S. Department of Labor, the Pension Benefit Guaranty Corporation (the "PBGC"), the Canada Revenue Agency, the applicable pension authority in Canada, or any participant or beneficiary with respect to any Employee Plan or any other employee benefit plan, agreement, commitment or arrangement in the past maintained by the Company or any of its Benefits Affiliates or to which the Company or any of its Benefits Affiliates has ever been under an obligation to contribute. Neither the Company nor, to the knowledge of the Company, any plan fiduciary of any Pension Plan or Welfare Plan has engaged in any transaction in violation in any material respect of Section 406(a) or (b) of ERISA (for which no exemption exists under Section 408 of ERISA) or any "prohibited transaction" (as defined in Section 4975(c)(1) of the Code) for which no exemption exists under Section 4975(c)(2) or 4975(d) of the Code.
- (e) Neither the Company nor any of its Benefits Affiliates has ever been a sponsor of, contributed to, or been under an obligation to contribute to any Pension Plan that is subject to Title IV of ERISA or Section 412 or Section 430 of the Code, any "multiemployer plan," as such term is defined in Section 3(37) of ERISA, or, in the case of an Employee Plan in Canada, any registered pension plan, as defined under the ITA, or a "pension plan" as defined under applicable pension benefits standards legislation.
- (f) Except as provided for in this Agreement, neither the Company nor any of its Benefits Affiliates is a party to any oral or written (i) agreement with any director, officer or other employee of the Company or any of its Benefits Affiliates the benefits of which are contingent, or the terms of which are materially altered, upon the occurrence of a transaction involving the Company of the nature contemplated by this Agreement; or (ii) agreement or plan, any of the benefits of which will be increased, or the vesting of the benefits of which will be accelerated, by the occurrence of any of the transactions

contemplated by this Agreement or the value of any of the benefits of which will be calculated on the basis of any of the transactions contemplated by this Agreement.

- (g) No Employee Plan provides health, dental, or life insurance benefits to any employee of the Company or any Benefits Affiliate, or any dependent of such an employee, following termination of the employee's employment, except as may be required by Section 4980B of the Code or any similar state law.
- (h) The Company and its Benefits Affiliates have at all times satisfied in all material respects all contribution obligations to all applicable government sponsored pension or welfare benefit programs in Canada including the Canada Pension Plan (CPP) or Quebec Pension Plan (QPP), the Old Age Security Act (OAS) program, unemployment insurance, workers' compensation, and the health or medical services plan of the applicable Province.
- (i) Each Employee Plan that is a "nonqualified deferred compensation plan" within the meaning of Section 409A(d)(1) of the Code (a "Nonqualified Deferred Compensation Plan") and any award thereunder, in each case that is subject to Section 409A of the Code, has been operated in compliance in all material respects with Section 409A of the Code since January 1, 2005, based upon a good faith, reasonable interpretation of (i) Section 409A of the Code and (ii)(A) the proposed and final Treasury Regulations issued thereunder and (B) Internal Revenue Service Notice 2005-1, all subsequent Internal Revenue Service Notices and other interim guidance on Section 409A of the Code.
- (j) All employee data necessary to administer each Employee Plan is in the possession of the Company or the Benefits Affiliates or their agents and is in a form that is sufficient for the proper administration of the Employee Plan in all material respects in accordance with its terms and all Applicable Laws, and such data is complete and correct in all material respects. No condition exists that would prevent the Company or its Benefits Affiliates from amending or terminating any Employee Plan, other than any limitations imposed by Applicable Law.
- (k) For purposes of this Agreement, the term "Benefits Affiliate" (i) with respect to any Employee Plan in the U.S., includes any business entity that is required to be aggregated and treated as one employer with the Company under Section 414(b), (c) or (m) of the Code, or (ii) with respect to any Employee Plan in Canada, any business entity with which the Company does not deal at arms length as determined under Section 251 of the ITA.

3.15 Employment and Labor Matters.

(a) There are no material outstanding, pending, or, to the knowledge of the Company, threatened or anticipated assessments, actions, causes of action, claims, complaints, demands, orders, prosecutions, or suits against the Company or its subsidiaries, or any of their respective directors, officers or agents, pursuant to or under

the Canada Pension Plan or any other Applicable Laws relating to unemployment insurance, employer health tax, employment standards, labor relations, occupational health and safety, human rights, workers' compensation, or pay equity. The Company has no obligation to reinstate any employees.

(b) The Company has not made any agreements, whether directly or indirectly, with any labor union, employee association, or other similar entity or made commitments to or conducted negotiations with any labor union or employee association or other similar entity with respect to any future agreements. No trade union, employee association, or other similar entity has any bargaining rights acquired by either certification or voluntary recognition with respect to the employees. None of the Company or any of its subsidiaries is aware of any current attempts to organize or establish any labor union, employee association, or other similar entity affecting the Company or its subsidiaries. There are no outstanding labor tribunal proceedings of any kind, including any proceedings that would be reasonably expected to result in certification of a trade union as bargaining agent for the employees, and there have not been any such proceedings. To the knowledge of the Company, there are no threatened or apparent union organizing activities involving the employees. To the knowledge of the Company or any of its subsidiaries has any material labor problems that would be reasonably expected to affect the value of the business or lead to any interruption of the Company's or its subsidiaries' operations at any location.

3.16 Environmental.

- (a) Neither the Company nor any of its subsidiaries has received written notice of, or, to the knowledge of the Company, is subject to, any pending or threatened action, cause of action, claim, or investigation alleging liability under or non-compliance with any Applicable Laws relating to pollution or the protection of human health or the environment ("Environmental Laws"), except for such actions, causes of action, claims, or investigations that would not reasonably be expected to have a Material Adverse Effect. The Company and each of its subsidiaries are in compliance with all Environmental Laws, except where the failure to so comply would not reasonably be expected to have a Material Adverse Effect. The Company and each of its subsidiaries holds and is in compliance with all permits required to be held by it under Environmental Laws, except where the failure to hold any such permit or to comply with any such permit would not reasonably be expected to have a Material Adverse Effect.
- (b) To the knowledge of the Company, there has been no spill, discharge, leak, emission, injection, disposal, escape, dumping, or release of any kind (collectively, "Release") of any pollutants, contaminants, hazardous substances, hazardous chemicals, toxic substances, hazardous wastes, infectious wastes, radioactive materials, materials, petroleum (including crude oil or any fraction thereof) or solid wastes, including those defined in any Environmental Law ("Hazardous Materials"), on, beneath, above, or into any of the Owned Real Property or the Leased Real Property, except for any Releases permitted by law or that have not had and would not reasonably be expected to have a Material Adverse Effect.

- (c) Except for any matters that have not had and would not reasonably be expected to have a Material Adverse Effect, there is no Environmental Claim pending or, to the knowledge of the Company, threatened against the Company or any of its subsidiaries or against any person or entity whose liability for such Environmental Claim the Company or any of its subsidiaries has retained or assumed either contractually or by operation of law. "Environmental Claim" means any notice by a person or entity alleging potential liability (including potential liability for investigatory costs, cleanup costs, governmental response costs, natural resources damages, property damages, personal injuries, or penalties) arising out of, based on, or resulting from (i) the presence, or release into the environment, of any material or form of energy at any location, whether or not owned by the Company or any of its subsidiaries; or (ii) circumstances forming the basis of any violation, or alleged violation, of any Environmental Law.
- (d) Neither the Company nor any of its subsidiaries has released any Hazardous Materials at or from any Owned Real Property or Leased Real Property, except as would not reasonably be expected to have a Material Adverse Effect.
- (e) The Company has made available to Parent complete and correct copies of all reports, licenses, permits, authorizations, disclosures, and other documents of which it is aware relating in any way to the status of any of the Owned Real Property or the Leased Real Property or otherwise relating to the business of the Company or any of its subsidiaries with respect to any Environmental Law.
- 3.17 Suppliers and Customers. Section 3.17 of the Disclosure Schedule lists the names of the 20 largest customers, and the 20 largest suppliers, of the Company and its subsidiaries, taken as a whole, for the 12-month period that commenced April 1, 2007 and ended on March 31, 2008. As of the date hereof, no such customer or supplier of the Company or any of its subsidiaries has canceled, or otherwise so modified in a manner materially adverse to the Company and its subsidiaries, taken as a whole, or given written notice to the Company or any of its subsidiaries of an intention to so cancel or otherwise so modify, its business relationship with the Company or any of its subsidiaries.
- 3.18 <u>Insurance</u>. Section 3.18 of the Disclosure Schedule lists (a) all material policies of liability, property, casualty, environmental, and other forms of insurance owned or held by the Company and each of its subsidiaries, copies of which have previously been made available to Parent; and (b) all material insurance claims filed by the Company under such policies that have not been paid in full as of the date hereof and the amounts claimed thereunder. All such insurance policies maintained by the Company and its subsidiaries are in full force and effect, all premiums required to have been paid with respect thereto have been paid, no notice of cancellation in respect thereof has been received and none of such insurance policies will terminate or lapse as a result of the consummation of the transactions contemplated by this Agreement.
- 3.19 Competition Act (Canada). For the purposes of determining the application of the pre-merger filing requirements of the Competition Act (Canada), the Company, together with

its affiliates (as defined in the Competition Act (Canada)), has assets in Canada with an aggregate value of approximately C\$5,615,000 and gross revenues from sales in, from, or into Canada with an aggregate value of approximately C\$12,925,000, as shown on the Balance Sheet and corresponding statement of operations.

- 3.20 Antitakeover Provisions. All necessary approvals have been granted by the Company Board under Section 203 of the Delaware Law so that none of the execution of the Voting Agreement, the granting of the Irrevocable Proxies, or any beneficial ownership of Company Common Stock by Parent, Operating Sub, Sub or any of Parent's other affiliates after the execution of this Agreement will limit, delay, or impair the consummation of the Merger or any other transaction with the Company or any of its subsidiaries by Parent, Operating Sub, Sub or any of Parent's other affiliates under Section 203 of the Delaware Law. No "fair price," "business combination," "moratorium," "control share acquisition," or other similar antitakeover statute, regulation, or charter provision applies or purports to apply to the Company with respect to this Agreement, the Merger, or any other transactions contemplated by this Agreement.
- **3.21** <u>Stockholder Voting Requirement</u>. The only vote of the Company's stockholders necessary to consummate the Merger under the Delaware Law and the Company's Certificate of Incorporation and Bylaws is the affirmative vote of the holders of a majority of the Company Common Stock.
- **3.22** <u>Associate Transactions</u>. Except as disclosed in the SEC Reports filed before the date of this Agreement, no Associate (a) purchases from or sells or furnishes to the Company or any of its subsidiaries any material goods or services; (b) owns, leases or licenses any real or material personal property that is used by the Company or any of its subsidiaries; or (c) is a party to any contract or agreement for joint purchases or sales of any material goods or services with the Company or any of its subsidiaries.
- 3.23 <u>Unregistered Sale of Securities</u>. The Company has neither issued nor sold any Company securities that were not registered at the time of sale under the United States Securities Act of 1933 (the "<u>Securities Act</u>") or otherwise issued pursuant to an available exemption from registration under the Securities Act.

3.24 Brokers; Finders; Transaction Fees.

(a) Except for the fees of RBC Dominion Securities Inc. as financial advisor to the Company, which fees the Company agrees to pay, there are no claims for brokerage commissions, finders' fees, investment advisory fees, or similar compensation in connection with this Agreement or the transactions contemplated by this Agreement, based on any arrangement, understanding, commitment, or agreement made by or on behalf of the Company or any of its subsidiaries, obligating the Company or any of its subsidiaries, or Parent, Operating Sub, or Sub, to pay such claim. The Company has made available to Parent a true and correct copy of all agreements under which any fees or other remuneration are due from the Company or any of its subsidiaries to RBC Dominion Securities Inc.

- (b) Section 3.24 of the Disclosure Schedule sets forth a statement of all (i) financial advisory fees (including fairness opinion costs), (ii) estimated legal fees and expenses (except for expenses relating to printing and mailing the Information Circular and any fees and expenses relating to any litigation relating to the Merger or the other transactions contemplated by this Agreement), or (iii) any other estimated professional fees and expenses, in each case arising out of or related to the completion of the Merger or the other transactions contemplated by this Agreement incurred or to be incurred by the Company or any of its subsidiaries through the Effective Time.
- 3.25 <u>Fairness Opinion</u>. The Company Board has received an opinion of RBC Dominion Securities Inc. to the effect that, as of the date of this Agreement, the Merger Consideration to be received in the Merger by the holders of Company Common Stock is fair to such holders from a financial point of view.

Article IV Representations and Warranties of Parent, Operating Sub, and Sub

Each of Parent, Operating Sub, and Sub, jointly and severally, represents and warrants to the Company as follows:

4.1 Corporate Organization. Parent is a corporation duly organized, validly existing, and in good standing under Minnesota law. Each of Operating Sub and Sub is a corporation duly organized, validly existing, and in good standing under the laws of the State of Delaware.

4.2 Authority; Non-Contravention.

- (a) Each of Parent, Operating Sub, and Sub has the corporate power to execute this Agreement and to consummate the transactions contemplated hereby. The execution, delivery, and performance of this Agreement by Parent, Operating Sub, and Sub have been duly and effectively authorized by the respective Boards of Directors of such corporations, and, subject to Section 5.11, no further corporate action is necessary on the part of Parent, Operating Sub, or Sub to authorize the consummation of the transactions contemplated hereby. This Agreement has been duly and validly executed and delivered by Parent, Operating Sub, and Sub and, assuming the accuracy of the representations and warranties of the Company set forth in the first two sentences of Section 3.3(a), constitutes a valid and binding agreement of Parent, Operating Sub, and Sub, and Sub, enforceable against Parent, Operating Sub, and Sub in accordance with its terms, except to the extent that enforceability may be limited by the Enforcement Limitations.
- (b) Neither the execution and delivery of this Agreement by Parent, Operating Sub, and Sub, nor the consummation by Parent, Operating Sub, or Sub of the transactions contemplated hereby, will (i) conflict with or result in a breach of the corporate charter or bylaws, as currently in effect, of Parent, Operating Sub, or Sub; (ii) require the consent or approval of, or any filing with, any Governmental Authority having jurisdiction over any

of the businesses or assets of Parent, Operating Sub, or Sub, or Sub, or violate any order, writ, injunction, decree, statute, rule, or regulation applicable to Parent, Operating Sub, or Sub or any of their properties or assets; or (iii) result in a breach of or constitute a default or an event that, with the passage of time or the giving of notice, or both, would constitute a default, or require notice to or the consent of any third party under, any other instrument, contract, or agreement to which Parent, Operating Sub, or Sub is a party or by which any of them or any of the properties or assets of any of them may be bound, except, in the case of clauses (ii) and (iii), (A) the filing of the Certificate of Merger with the Secretary of State of the State of Delaware; and (B) where such violations, breaches, or defaults, or the failure to obtain, make or give such consents, approvals, filings, or notices, would not, individually or in the aggregate, be reasonably expected to impair Parent's, Operating Sub's, or Sub's ability to consummate the Merger or the other transactions contemplated hereby.

- (c) No antitrust or competition-law notices, filings, or approvals by Parent, Operating Sub, or Sub, including those under the United States Hart-Scott-Rodino Antitrust Improvements Act of 1976, the Investment Canada Act, or the Competition Act (Canada), are required in connection with the Merger.
- 4.3 Adequate Funds. Parent, Operating Sub, and Sub have, and will have at the Effective Time, sufficient funds for the payment of the aggregate Merger Consideration.
- 4.4 <u>Litigation</u>. There are no claims, litigation, arbitrations, administrative proceedings, abatement orders, or investigations of any kind pending or, to the knowledge of Parent, threatened against Parent, Operating Sub, or Sub, which, if decided adversely to Parent, Operating Sub, or Sub, are reasonably likely to prevent, materially delay, or materially impair Parent's or Sub's ability to consummate the transactions contemplated by this Agreement. There are no judgments, orders, writs, injunctions, decrees, indictments, subpoenas or civil investigative demands or awards against Parent, Operating Sub, or Sub that are reasonably likely to prevent, materially delay, or materially impair Parent's or Sub's ability to consummate the transactions contemplated by this Agreement.
- 4.5 <u>Finders; Brokers</u>. Except for fees of Cherry Tree & Associates, LLC, which Parent agrees to pay, there are no claims for brokerage commissions, finders' fees, investment advisory fees, or similar compensation in connection with this Agreement or the transactions contemplated by this Agreement, based on any arrangement, understanding, commitment, or agreement made by or on behalf of Parent, Operating Sub, or Sub, obligating the Company or any of its subsidiaries, or Parent, Operating Sub, or Sub, to pay such claim.

Article V Pre-Closing Covenants

5.1 Operation of Business of the Company. From the date of this Agreement through the earlier of termination of this Agreement or the Effective Time, and except as set forth in Section 5.1 of the Disclosure Schedule:

- (a) The Company shall use its reasonable best efforts to preserve intact in all material respects its business organization, assets, and technology and those of its subsidiaries, to maintain its rights and franchises and those of its subsidiaries, to keep available to itself and to the Surviving Corporation the services of the present officers and employees of the Company and its subsidiaries, and to preserve for itself and for the Surviving Corporation the present relationships of the Company and its subsidiaries with persons having significant business dealings with the Company or any of its subsidiaries.
- (b) The Company shall, and shall cause each of its subsidiaries to, except as otherwise consented to in writing by Parent, conduct its business and operations in the ordinary course consistent with past practice.
- (c) Unless otherwise required by the terms of this Agreement, the Company shall not, and shall not permit any of its subsidiaries to, except as otherwise consented to in writing by Parent:
- (i) amend its Certificate of Incorporation, Bylaws, or other comparable charter or organizational documents, or alter through merger, liquidation, reorganization, restructuring, or in any other fashion the corporate structure or ownership of any subsidiary of the Company;
 - (ii) increase or decrease the number of authorized shares of its capital stock;
- (iii) issue, grant, sell, or pledge any shares of capital stock or other equity securities of the Company or any of its subsidiaries (other than the issuance of shares of Company Common Stock upon exercise of Options outstanding as of the date of this Agreement that are disclosed in Section 3.2 in accordance with their terms) or any options, warrants, or other rights to purchase any such capital stock or other equity securities or any securities or any securities or any stock appreciation rights, performance shares, phantom stock, or other similar rights based upon the value of any such capital stock or other equity securities, or reprice any Options;
 - (iv) split, combine, or reclassify any shares of its capital stock or make any other changes in its equity capital structure;
- (v) purchase, redeem, or cancel for value, or permit any of its subsidiaries to purchase, redeem, or cancel for value, directly or indirectly, any shares of capital stock or other equity securities of the Company or any of its subsidiaries or any Options or other rights to purchase any such capital stock or other equity securities or any securities convertible into or exchangeable for any such capital stock or other equity securities, except as contemplated by Section 1.8;

- (vi) declare, set aside, or pay, or permit any of its subsidiaries to declare, set aside, or pay, any dividend or other distribution or payment in cash, stock, or property in respect of shares of its capital stock or other equity securities, except that any subsidiary of the Company may pay dividends or other distributions to the Company or any of its other wholly owned subsidiaries;
 - (vii) designate any class or series of shares of Company Preferred Stock;
- (viii) purchase, lease, or otherwise acquire (including acquisitions by merger, consolidation, or stock or asset purchase) any assets or properties, other than those the fair value of which does not exceed C\$100,000 individually or C\$250,000 in the aggregate, and other than inventory and supplies acquired in the ordinary course of business consistent with past practice;
- (ix) sell, lease, encumber, mortgage, or otherwise dispose of any material assets or properties, except that the Company and its subsidiaries may sell or otherwise dispose of inventory and obsolete equipment in the ordinary course of business consistent with past practice;
- (x) waive, release, grant, or transfer any rights of material value or modify or change in any material respect any existing license, contract, or other document or agreement, other than in the ordinary course of business consistent with past practice;
- (xi) incur any indebtedness for money borrowed, other than indebtedness of the Company to its wholly owned subsidiaries or of a wholly owned subsidiaries or of a wholly owned subsidiaries, or incur any purchase money indebtedness for fixed assets or enter into any financing, "synthetic," or capitalized lease;
- (xii) incur any other liability or obligation (except of the Company to its wholly owned subsidiaries or of a wholly owned subsidiary to the Company or its other wholly owned subsidiaries), other than in the ordinary course of business consistent with past practice, or assume, guarantee, endorse (other than endorsements of checks in the ordinary course of business) or otherwise as an accommodation become responsible for the obligations of any other person (except by the Company with respect to the obligations of its wholly owned subsidiaries or by a subsidiary with respect to the obligations of the Company or its other wholly owned subsidiaries);
- (xiii) except as otherwise required by this Agreement, (A) enter into any new employee benefit plan, program, or arrangement, or any new employment, severance, or consulting agreement; (B) amend any existing employee benefit plan, program, or arrangement, or any existing employment, severance, or consulting agreement; (C) pay any retention, "stay," transaction, or other bonuses

in connection with the transactions contemplated by this Agreement; (D) grant any increases in compensation or benefits other than pursuant to customary salary and employee benefit administration in the ordinary course of business consistent with past practice; (E) hire any new employee of the Company or any of its subsidiaries at an annual compensation rate in excess of customary practice or, in any event, in excess of C\$100,000; or (F) engage any new consultant to the Company or any of its subsidiaries outside of the ordinary course of business;

- (xiv) enter into, extend, renew, modify, or amend any collective bargaining agreement;
- (xv) enter into any other material transaction, other than in the ordinary course of business consistent with past practices;
- (xvi) make any tax election or settle or compromise any material tax liability;
- (xvii) change any accounting principles used by it, unless required by GAAP;
- (xviii) settle any litigation, proceedings, or material claims other than those arising in the ordinary course of business;
- (xix) enter into any agreement with any affiliate of the Company or any Associate, other than agreements solely between the Company and one or more of its wholly owned subsidiaries or between two or more of the Company's wholly owned subsidiaries; or
 - (xx) enter into any contract, agreement, commitment, or arrangement with respect to any of the foregoing.

5.2 Stockholders' Meeting; Information Circular.

(a) The Company shall cause a special meeting of its stockholders to be duly called and held as soon as reasonably practicable after the execution of this Agreement for the purpose of voting on the adoption and approval of this Agreement and the Merger. Subject to Section 5.2(e), the Information Circular (as defined in Section 5.2(b)) shall contain the unanimous recommendation of the Company Board to the stockholders of the Company that they vote in favor of adoption and approval of this Agreement and the Merger, the Company shall solicit proxies in connection with the meeting in favor of such adoption and approval, and the Company shall otherwise use its reasonable best efforts to secure the adoption and approval of the stockholders of the Company required to effect the Merger under Applicable Law and the Company's Certificate of Incorporation and Bylaws. The Company's obligations to call and hold the stockholders' meeting contemplated by this Section 5.2(a) shall not be affected by the announcement of, or the Company's receipt of, an Acquisition Proposal (as defined in Section 5.3(a)) or

by the failure to make a recommendation or any withdrawal, qualification, or adverse modification of the Company Board's approval and recommendation of this Agreement and the Merger (or any public proposal to do any of the foregoing) (an "Adverse Recommendation Change"), provided that nothing stated in this Section 5.2(a) limits the Company's right to terminate this Agreement under Section 8.1(e).

- (b) Contemporaneously with the first public announcement by the parties of the execution of this Agreement, the Company shall publish a notice of the date of the stockholders' meeting described in Section 5.2(a), which date shall be a sufficient period of time after such notice to comply with Applicable Laws (but in no event will the date specified in the initial publication of such notice be later than July 31, 2008). The Company shall prepare an information circular, together with a form of proxy, with respect to the stockholders' meeting as soon as reasonably practicable after the execution of this Agreement (such information circular, together with any amendments thereof or supplements thereto, being the "Information Circular"). The Information Circular shall comply in all material respects with the requirements of Applicable Law. The Company shall (i) as soon as reasonably practicable after the date hereof, mail the Information Circular to the stockholders of the Company and (ii) otherwise comply in all material respects with all other applicable legal requirements in respect of such meeting. Before mailing the Information Circular, the Company shall provide a reasonable opportunity for Parent to review and comment upon the contents of the Information Circular and shall not include therein any information to which Parent reasonably objects (provided that the Company shall retain the right to make any final decisions with respect to the exclusion of any information from the Information Circular) or omit therefrom any information that Parent reasonably requests. If, at any time before the stockholders' meeting, any event relating to the Company or any of its subsidiaries, officers, or directors is discovered by the Company that should be set forth in an amendment or supplement to the Information Circular, the Company shall promptly so inform Parent.
- (c) The Company covenants that the Information Circular will not, at the time the Information Circular is mailed to the Company's stockholders, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading, and will not, at the time of the meeting of stockholders to which the Information Circular relates, omit to state any material fact necessary to correct any statement that has become false or misleading in any earlier communication with respect to such meeting, except that no covenant is made by the Company with respect to statements made in the Information Circular based on information furnished in writing to the Company by Parent, Operating Sub, or Sub specifically for use in the Information Circular. The Company covenants that the Information Circular will comply as to form and substance in all material respects with the requirements of the Canadian securities rules (National Instrument 51-102) and any other Applicable Laws.
- (d) Parent covenants that none of the information supplied or to be supplied in writing by Parent, Operating Sub, or Sub specifically for inclusion in the Information Circular will, at the time the Information Circular is mailed to the Company's

stockholders, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading, or will, at the time of the meeting of stockholders to which the Information Circular relates, as then amended or supplemented, omit to state any material fact necessary to correct any statement that has become false or misleading in any earlier communication with respect to such meeting.

- (e) The Company Board may make an Adverse Recommendation Change only if it determines in good faith (after consultation with the Company's outside legal counsel) that it is required to do so in order to comply with its fiduciary duties under Delaware law.
- (f) The Company may not submit to the vote of its stockholders any Acquisition Proposal or any proposal for a Third-Party Transaction (as defined in Section 5.3(e)) before the termination of this Agreement.

5.3 No Solicitation.

- (a) From the date of this Agreement until the earlier of the Effective Time or the termination of this Agreement, the Company shall not, and shall not authorize or permit any of its officers, directors, employees, financial advisers, legal counsel, representatives, agents, subsidiaries, or affiliates to, directly or indirectly:
 - (i) solicit, seek, initiate, or encourage any inquiries or proposals that constitute, or would be reasonably likely to lead to, a proposal or offer for a merger, consolidation, amalgamation, arrangement, or other business combination involving the Company or any of its subsidiaries, a sale of substantial assets of the Company and its subsidiaries, taken as whole (other than the sale or other disposition of inventory or obsolete equipment in the ordinary course of business consistent with past practice), a sale of shares of capital stock of the Company or any of its subsidiaries (including by way of a tender offer or takeover bid), or any similar transaction involving the Company or any of its subsidiaries, other than the transactions contemplated by this Agreement (any of the foregoing inquiries or proposals being an "Acquisition Proposal");
 - (ii) engage in discussions or negotiations with any person or group other than Parent or its affiliates (a "Third Party") concerning any Acquisition Proposal, or provide any non-public information, or afford access to the properties, books, records, or personnel of the Company or any of its subsidiaries, to any Third Party that is considering making, or has made, any Acquisition Proposal;
 - (iii) enter into any letter of intent, agreement in principle, or other agreement, arrangement, or understanding with respect to an Acquisition Proposal; or

- (iv) otherwise agree to or recommend any Acquisition Proposal.
- (b) Notwithstanding anything to the contrary in Section 5.3(a), nothing contained in Section 5.3(a) prevents the Company from:
- (i) before obtaining the adoption and approval of this Agreement and the Merger by the Company stockholders as contemplated under Section 5.2(a), furnishing non-public information or affording access to the properties, books, records, or personnel of the Company or any of its subsidiaries to, or entering into discussions or negotiations with, any Third Party in connection with an Acquisition Proposal if and only to the extent that:
 - (A) the Acquisition Proposal was made after the date of this Agreement and was not solicited, sought, encouraged, or initiated in violation of this Agreement;
 - (B) the Company Board, in the exercise of its fiduciary duties, determines in good faith (after consultation with the Company's financial adviser and outside legal counsel) that the Acquisition Proposal is, or is reasonably likely to result in, a Superior Proposal; and
 - (C) before furnishing such non-public information, affording such access, or entering into such discussions or negotiations, the Company receives from the Third Party an executed confidentiality agreement with terms no less favorable to the Company in all material respects than those contained in the letter agreement regarding confidentiality dated March 13, 2008 from the Company to Parent (the "Confidentiality Agreement") and the Company contemporaneously furnishes such information to Parent (to the extent such information has not been previously furnished by the Company to Parent); or
 - (ii) complying with Rule 14e-2 promulgated under the Exchange Act with regard to an Acquisition Proposal.
- (c) The Company shall, and shall cause its officers, directors, employees, financial advisers, legal counsel, representatives, agents, subsidiaries, and affiliates to, immediately cease and cause to be terminated immediately all existing activities, discussions, and negotiations with any Third Parties conducted heretofore with respect to, or that would reasonably be expected to lead to, any Third-Party Transaction.
- (d) The Company will not release any Third Party from, or waive any provision of, any confidentiality or standstill agreement to which it is a party, unless the Company Board determines in good faith (after consultation with the Company's

financial adviser and outside legal counsel) that such action is required in order for the Company Board to comply with its fiduciary duties under Delaware law.

- (e) For purposes of this Agreement:
- (i) "Superior Proposal" means a bona fide written proposal made by a Third Party for a Third-Party Transaction (provided that for purposes of this definition, references to "10% or more" or "25% or more" in the definition of "Third-Party Transaction" will be deemed references to "a majority") that was not solicited, sought, initiated, or encouraged by the Company in violation of this Agreement, and that, in the good faith judgment of the Company Board (after consulting with the Company's financial adviser and outside legal counsel), taking into account, to the extent deemed appropriate by the Company Board in the exercise of its fiduciary duties, the various legal, financial, regulatory, and other aspects of the Acquisition Proposal and the person or group making such proposal (including the absence of committed financing to the extent financing is a condition to the consummation of the Third-Party Transaction and other conditions to consummation), (A) if accepted, is reasonably likely to be consummated, and (B) if consummated, would result in a transaction that is more favorable to the Company's stockholders (in their capacity as stockholders) from a financial point of view than the Merger (taking into account any proposed adjustments made by Parent under Section 5.3(f)(ii)).
- (ii) "Third-Party Transaction" means (A) an acquisition after the date of this Agreement under an Acquisition Proposal, except that (1) with respect to equity securities, it means an acquisition of equity securities of the Company that, when aggregated with all other equity securities of the Company beneficially owned (as defined in Rule 13d-3 promulgated under the Exchange Act), immediately after entering into the definitive agreement for, or consummation of, the Third-Party Transaction, by the Third Party making such Acquisition Proposal or any of its affiliates, constitutes 10% or more of the total equity interests in, and 10% or more of the total voting power of the then-outstanding equity securities of, the Company, and (2) with respect to an acquisition of assets, it means an acquisition of assets of the Company or any of its subsidiaries constituting 25% or more, on a fair-market-value basis, of the total assets of the Company and its subsidiaries on a consolidated basis; (B) the adoption by the Company of a plan of liquidation or dissolution; (C) the repurchase of, or recapitalization involving, 25% or more of the Company's outstanding equity securities; or (D) the payment of an extraordinary dividend or other distribution on Company Common Stock equal to 25% or more of the Company Common Stock's then-current market price.
- (f) If the Company Board determines in good faith in the exercise of its fiduciary duties (after consultation with the Company's financial adviser and outside legal counsel) to recommend or approve a Superior Proposal if Parent does not take the action it is entitled to take under Section 5.3(f)(ii), then:

- (i) the Company shall, at least four business days prior thereto, give Parent written notice thereof (which notice shall not constitute an Adverse Recommendation Change) and furnish Parent with a copy of the definitive agreement the Company is prepared to execute with respect to the transactions contemplated by the Superior Proposal and shall afford a reasonable opportunity to Parent within such four-business-day period to make such adjustments to this Agreement as would enable the Company Board to maintain its recommendation of this Agreement and the Merger to the stockholders of the Company and enable the Company to proceed with the Merger on such adjusted terms (it being understood and agreed that any amendment to the financial terms or any other material term of the Superior Proposal will require a new four-business-day notice and an additional right of the Parent to make adjustments to this Agreement); and
- (ii) the Superior Proposal shall not be recommended or approved by the Company Board nor shall any notice of termination of this Agreement be given by the Company under Section 8.1(e) in connection therewith if Parent submits to the Company during such four-business-day period a legally binding, executed offer to enter into an amendment to this Agreement within such four- business-day period reflecting such adjustments, unless the Company Board shall have determined in good faith (after consultation with the Company's financial adviser and outside legal counsel) that the transactions contemplated herein, as modified by the amendment to this Agreement that Parent has agreed to enter into during such four-business-day period, would not, if consummated, result in a transaction that is at least as favorable to the Company's stockholders (in their capacity as stockholders) from a financial point of view as the Superior Proposal.
- (g) Notwithstanding anything to the contrary stated herein, the Company Board may recommend, approve, or accept a Superior Proposal only if the Company concurrently therewith terminates this Agreement under Section 8.1(e) and, concurrently with such termination, makes the payments required by Section 8.3.
- (h) The Company shall notify Parent immediately after receipt by the Company (or its advisers) of any Acquisition Proposal or any request for non-public information, or for access to the properties, books, records, or personnel of the Company or any of its subsidiaries, by any Third Party that is considering making, or has made, an Acquisition Proposal. The notice shall be made orally and in writing and shall indicate in reasonable detail the terms and conditions of the Acquisition Proposal or request (including the identity of the Third Party making the Acquisition Proposal or request). The Company shall continue to keep Parent informed, on a reasonably current basis, of the status of any discussions or negotiations regarding any Acquisition Proposal and the terms being discussed or negotiated (including changes or amendments thereto).
 - (i) The Company agrees that any violations of the restrictions set forth in this Section 5.3 by any of its or its subsidiaries' respective officers, directors, advisers,

consultants, or employees will be deemed to be a breach of this Section 5.3 by the Company.

5.4 Access to Information. The Company will give Parent, and its counsel, financial advisers, auditors, and other authorized representatives, full access to the offices, properties, personnel, books, and records of the Company and its subsidiaries at all reasonable times upon reasonable notice, and will instruct the personnel, counsel, financial advisers, and auditors of the Company and its subsidiaries to cooperate in all reasonable respects with Parent and each such representative in its investigation of the business of the Company and its subsidiaries, provided that no investigation under this Section 5.4 will affect any representation or warranty given by the Company to Parent, Operating Sub, and Sub hereunder or unreasonably interfere with the operations of the Company and its subsidiaries. The Company will confer from time to time with Parent at Parent's request to discuss the status of the operations of the Company and its subsidiaries.

5.5 Employee Benefit Plans.

- (a) The Company will cause any Employee Plans that it may maintain to be amended for the purpose of permitting the Employee Plans to continue to operate in conformity with ERISA, the Code, the ITA, the pension standards legislation of Canada or any applicable province and other Applicable Laws in effect on or before the Merger, subject to approval of any such amendment by Parent. The Company will take any actions necessary to terminate effective at or, at the election of Parent, immediately before the Effective Time, any Employee Plans that Parent requests to be terminated.
- (b) Parent shall take such actions as are necessary to provide each employee of the Company and its subsidiaries with credit for service for the Company and its Benefits Affiliates for purposes of vesting, eligibility, and participation (but not benefit accruals under any defined benefit plan or eligibility for post-retirement medical benefits) under any benefit plan or arrangement of Parent or an affiliate thereof in which such employee of the Company and its subsidiaries may participate on or after the Effective Time in the same manner as if such service had been service for the Parent or its affiliate; provided, however, that (i) service shall only be credited to the extent such service was recognized by the Company or its Benefits Affiliates prior to the Effective Time, and (ii) no such credit shall be required to the extent that such credit would result in a duplication of benefits for the same period of service.
- (c) Subject to the requirements of Applicable Law, Parent shall take such actions as are necessary to cause the group health plan maintained by Parent or an affiliate thereof, and applicable insurance carriers, third-party administrators, and any other third parties, to the extent such group health plan is made available to employees of the Company and its subsidiaries, to (i) waive any evidence-of-insurability requirements, waiting periods, and limitations as to preexisting medical conditions under the group health plan that are applicable to employees of the Company and its subsidiaries and their spouses and eligible dependents (but only to the extent that such requirements, waiting periods, and preexisting condition limitations did not apply or were satisfied under the

group health plan maintained by the Company or its Benefits Affiliates before the Effective Time) and (ii) provide each employee of the Company or any of its subsidiaries with credit, for the calendar year in which the Effective Time occurs, for the amount of any out-of-pocket expenses that are paid by such employee before the Effective Time under the group health plan maintained by the Company or its Benefits Affiliates during such calendar year toward any annual deductible or annual out-of-pocket maximum that applies under the group health plan maintained by Parent or any of its affiliates that may be extended to such employee during such calendar year.

- (d) If Parent directs pursuant to Section 5.5(a) that an Employee Plan qualified under Section 401(a) of the Code be terminated before the Effective Time, Parent shall take such actions as are necessary to cause a retirement plan maintained by it or one of its affiliates that is qualified under Section 401(a) of the Code to accept direct and indirect rollover distributions from such Employee Plan, by the employees of the Company and its subsidiaries who continue employment following the Closing, of account balances maintained by them under such Employee Plan, including promissory notes evidencing outstanding plan loans (if any).
- (e) This Section 5.5 is not intended to amend any benefit plans or arrangements of Parent or any of its affiliates, or to confer third-party beneficiary rights on any person who is not a party to this Agreement.
- 5.6 Reasonable Best Efforts. Subject to the terms and conditions of this Agreement, each of the parties hereto agrees to use its reasonable best efforts consistent with Applicable Law to take, or cause to be taken, all actions, and to do, or cause to be done, all things necessary or proper and advisable to ensure that the conditions set forth in Article VII are satisfied and to consummate and make effective, in the most expeditious manner practicable, the Merger and the other transactions contemplated by this Agreement.
- 5.7 Consents. Each of the parties hereto shall use its reasonable best efforts to obtain all material consents of third parties and Governmental Authorities, and to make all governmental filings, necessary for the consummation of the transactions contemplated by this Agreement.
- **5.8** <u>Public Announcements.</u> Parent and the Company will consult with each other before issuing any press release or otherwise making any public statements before the Effective Time with respect to the Merger or the other transactions contemplated hereby and shall not issue any such press release or make any such public statement before receiving the consent of the other party. Nothing in this Section 5.8 prohibits any party from making a press release or other statement required by Applicable Law or by obligations under any listing agreement with any stock exchange if the party making the disclosure has first consulted with the other parties hereto.
- 5.9 Notification of Certain Matters. The Company will give prompt notice, as soon as practicable, to Parent of the occurrence or nonoccurrence of any event, circumstance or condition that has (a) had or would reasonably be expected to have a Material Adverse Effect;

(b) caused any representation or warranty of the Company contained in this Agreement to be untrue or inaccurate in any material respect; or (c) caused any failure of the Company to comply in all material respects with or satisfy in all material respects any covenant, condition, or agreement to be complied with or satisfied by it under this Agreement; provided, however, that the delivery of any notice under this Section 5.9 will not limit or otherwise affect the rights or remedies of Parent, Operating Sub, or Sub under this Agreement.

5.10 <u>Approval of Sub's Sole Stockholder.</u> Immediately following the execution and delivery of this Agreement, Operating Sub, in its capacity as the sole stockholder of Sub, will approve and adopt this Agreement and the Merger.

Article VI Other Covenants

6.1 <u>Indemnification</u>. Parent shall cause the Surviving Corporation to maintain all rights to indemnification, expense advancement, and exculpation existing in favor of any person who is or was a director or officer of the Company or any of its subsidiaries before the Effective Time (collectively, the "<u>Indemnified Parties</u>"), when acting in such capacity or when acting as a fiduciary at the request of the Company under or with respect to an employee plan, as provided in the Company's Certificate of Incorporation or Bylaws or the certificate or articles of incorporation, bylaws, or similar organizational documents of any of its subsidiaries as in effect on the date hereof, and such provisions shall survive the Merger for a period of six years after the Effective Time (or, with respect to any relevant claim made within such six-year period, until final disposition of such claim) with respect to matters occurring at or before the Effective Time, and no action taken during such period shall be deemed to diminish the obligations set forth in this Section 6.1.

6.2 D&O Liability Insurance.

(a) For a period of six years after the Effective Time, Parent shall cause the Surviving Corporation to cause to be maintained in effect either (a) the current policy of directors' and officers' liability insurance maintained by the Company (provided that Parent or the Surviving Corporation may substitute therefor policies of at least the same coverage and amounts containing terms and conditions that are no less advantageous in any material respects to the Indemnified Parties thereunder) covering the Indemnified Parties with respect to claims arising from facts or events that occurred at or before the Effective Time; provided, however, that in no event will the Surviving Corporation be required to expend under this Section 6.2 more than an amount per year equal to 200% of the current annual premium paid by the Company for such existing insurance coverage (the "D&O Cap"); and provided, further, that if equivalent coverage cannot be obtained, or can be obtained only by paying an annual premium in excess of the D&O Cap, the Surviving Corporation shall only be required to obtain as much coverage as can be obtained by paying an annual premium equal to the D&O Cap, or (b) a run-off policy or endorsement with respect to the current policy of directors' and officers' liability insurance maintained by the Company covering the Indemnified Parties with respect to

claims asserted within six years after the Effective Time arising from facts or events that occurred at or before the Effective Time.

- (b) If the Surviving Corporation or any of its successors or assigns (i) consolidates with or merges into any other entity and shall not be the continuing or surviving corporation or entity of such consolidation or merger, or (ii) transfers or conveys all or substantially all of its properties or assets to any entity, then, in each such case, to the extent necessary, proper provision shall be made so that the successors and assigns of the Surviving Corporation shall assume the obligations set forth in Section 6.1 and this Section 6.2.
- (c) The rights of each Indemnified Party under Section 6.1 and this Section 6.2 shall survive the consummation of the Merger and are intended to benefit, and shall be enforceable by, each Indemnified Party.

Article VII Conditions

- 7.1 Conditions to the Obligations of Parent, Operating Sub, and Sub. The obligations of Parent, Operating Sub, and Sub to effect the Merger are subject to the fulfillment on or before the Closing Date of the following conditions, any one or more of which may be waived in writing by Parent, to the extent permitted by Applicable Law.
 - (a) Representations, Warranties, Covenants.
 - (i) The representations and warranties of the Company contained in Article III shall be true and correct both when made and on and as of the Closing Date as though made on and as of the Closing Date (except representations and warranties expressly made as of an earlier date, in which case as of such date), except to the extent that any inaccuracies in any such representations and warranties (other than the representations and warranties in Sections 3.2(a) and (b), which shall be deemed untrue and incorrect if not true and correct except to a *de minimis* extent when made or at any time prior to the Closing Date) have not had and would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect (provided that, solely for purposes of this exception, any representation or warranty in Article III that is qualified by materiality or Material Adverse Effect language shall be read as if such qualifier were not present).
 - (ii) The Company shall have performed and complied in all material respects with the agreements and obligations contained in this Agreement required to be performed and complied with by it at or before the Closing Date.
 - (iii) Parent, Operating Sub, and Sub shall have received a certificate signed on behalf of the Company by the Chief Executive Officer and the Chief

Financial Officer of the Company certifying that the conditions set forth in this Section 7.1(a) and in Section 7.1(d) have been satisfied.

- (b) <u>Stockholder Approval</u>. This Agreement and the Merger shall have been adopted and approved by the stockholders of the Company by the vote required by the Delaware Law and the Company's Certificate of Incorporation and Bylaws.
- (c) <u>Absence of Litigation</u>. There shall not be instituted or pending any suit, action, investigation, inquiry, or other proceeding brought by any Governmental Authority that is reasonably likely to restrain or prohibit the consummation of the transactions contemplated hereby or require rescission of this Agreement or such transactions or result in material damages, directly or indirectly, to Parent or the Surviving Corporation if the transactions contemplated hereby are consummated, and there shall not be in effect any injunction, preliminary restraining order, or other writ, order, judgment, or decree of any nature issued by a court or Governmental Authority of competent jurisdiction directing that any of the transactions contemplated hereby not be consummated or any statute, rule, or regulation enacted or promulgated that makes consummation of any of the transactions contemplated hereby illegal.

(d) Absence of Material Adverse Effect.

- (i) There shall not have been, since the date of this Agreement, any event, development, or condition (financial or otherwise) of any character that has had or would reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.
- (ii) For purposes of this Agreement, "Material Adverse Effect" means any change, effect, event, occurrence, state of facts, or development that, individually or in the aggregate with all other changes, effects, events, occurrences, states of facts or developments, (A) is materially adverse to the business, operations, results of operations, properties, assets, liabilities, or condition (financial or otherwise) of the Company and its subsidiaries, taken as a whole; provided, however, that none of the following will be taken into account in determining whether there has been or would reasonably be expected to be a Material Adverse Effect: (1) any change in the market price or trading volume of the Company Common Stock after the date hereof; (2) any change generally relating to the economy or securities markets of the United States or Canada or generally affecting the industry in which the Company and its subsidiaries operate that does not have a materially disproportionate effect on the Company and its subsidiaries; (3) any acts of God, terrorism or war (whether or not declared) that do not have a materially disproportionate effect on the Company and its subsidiaries; (4) any adverse change resulting from compliance with the terms of, or the taking of any action at the direction or Parent or required by, this Agreement, including the failure to take any action as a result of restrictions or other prohibitions set forth in Section 5.1 hereof; (5) any changes (after the date hereof) in GAAP or Applicable Law, except to the extent that any such change

has a materially disproportionate effect on the Company and its subsidiaries compared to other companies in the industry in which the Company and its subsidiaries operate; or (6) the announcement, pendency or consummation of the Merger (including any loss of or adverse change in the relationship of the Company and its subsidiaries with their respective employees, customers, partners or suppliers related thereto); or (B) would prevent the consummation of the Merger or any of the transactions contemplated hereby.

- (e) <u>Dissenting Shares</u>. The holders of not more than 10% of the issued and outstanding shares of Company Common Stock shall have taken such action before or at the time of the stockholders' vote on this Agreement and the Merger as is necessary as of that time to entitle them to the statutory dissenters' rights referred to in Section 1.7.
- (f) <u>Lien Release</u>. The Company shall provide a UCC-3 Termination Statement or other evidence reasonably satisfactory to Parent that the liens on the Company's assets in favor of The Royal Bank of Canada have been released.
- (g) Governmental and Regulatory Approvals. All consents, approvals and actions of, filings with, and notices to, any Governmental Authority, required of Parent, Operating Sub, Sub, the Company, or any of the Company's subsidiaries to consummate the Merger, the failure of which to be obtained or taken, individually or in the aggregate, is reasonably expected to materially impair the ability of the parties to consummate the Merger, shall have been obtained or taken, as the case may be.
- **7.2** Conditions to the Obligation of the Company. The obligation of the Company to effect the Merger is subject to the fulfillment at or before the Closing Date of the following conditions, any one or more of which may be waived in writing by the Company, to the extent permitted by Applicable Law.

(a) Representations, Warranties, and Covenants.

- (i) The representations and warranties of Parent, Operating Sub, and Sub contained in Article IV shall be true and correct both when made and on and as of the Closing Date as though made on and as of the Closing Date (except representations and warranties expressly made as of an earlier date, in which case as of such date), except to the extent that any inaccuracies in any such representations and warranties are not reasonably likely to impair, individually or in the aggregate, the consummation of the transactions contemplated by this Agreement.
- (ii) Each of Parent, Operating Sub, and Sub shall have performed and complied in all material respects with the agreements and obligations contained in this Agreement required to be performed and complied with by it at or before the Closing Date.

- (iii) The Company shall have received a certificate signed on behalf of Parent by an appropriate executive officer of Parent certifying that the conditions set forth in this Section 7.2(a) have been satisfied.
- (b) Stockholder Approval of Agreement and Merger. This Agreement and the Merger shall have been adopted and approved by the stockholders of the Company by the vote required by the Delaware Law and the Company's Certificate of Incorporation and Bylaws.
- (c) <u>Absence of Litigation, Injunctions</u>. There shall not be instituted or pending any suit, action, investigation, inquiry, or other proceeding brought by any Governmental Authority that is reasonably likely to restrain or prohibit the consummation of the transactions contemplated hereby or require rescission of this Agreement or such transactions, and there shall not be in effect any injunction, preliminary restraining order, or other writ, order, judgment, or decree of any nature issued by a court or Governmental Authority of competent jurisdiction directing that any of the transactions contemplated hereby not be consummated or any statute, rule, or regulation enacted or promulgated that makes consummation of any of the transactions contemplated hereby illegal.
- (d) Governmental and Regulatory Approvals. All consents, approvals and actions of, filings with, and notices to, any Governmental Authority required of Parent, Operating Sub, Sub, the Company or any of the Company's subsidiaries to consummate the Merger, the failure of which to be obtained or taken, individually or in the aggregate, is reasonably expected to materially impair the ability of the parties to consummate the Merger, shall have been obtained or taken, as the case may be.

Article VIII Termination

- **8.1** <u>Termination</u>. This Agreement may be terminated and the Merger and the other transactions contemplated hereby may be abandoned at any time before the Effective Time whether before or, except as otherwise provided in Section 8.1(e), after adoption and approval of this Agreement and the Merger by the stockholders of the Company:
 - (a) by mutual written consent of Parent and the Company;
 - (b) by Parent or the Company, (i) if the Merger shall not have been consummated on or before November 14, 2008, unless such failure of consummation is due to failure by the party seeking to terminate this Agreement (or, if Parent is seeking termination, the failure by either Parent, Operating Sub, or Sub) to comply in all material respects with, or the breach by such party (or, if Parent is seeking termination, the breach by either Parent, Operating Sub, or Sub) of, the terms, provisions, covenants, and agreements contained in this Agreement or (ii) if any Governmental Authority of competent jurisdiction shall have issued an order, decree, injunction or ruling or taken any other action permanently enjoining, restraining or otherwise prohibiting the

consummation of the Merger and such order, decree, ruling or other action shall have become final and nonappealable, or if there shall be adopted any law or regulation that makes consummation of the Merger illegal or otherwise prohibited;

- (c) (i) by Parent, if any of the conditions in Section 7.1 becomes impossible to fulfill, other than for reasons totally within the control of Parent, Operating Sub, or Sub, and shall not have been waived in writing by Parent, or (ii) by the Company or Parent, if the stockholders of the Company fail to adopt and approve this Agreement and the Merger by the vote required by the Delaware Law and the Company's Certificate of Incorporation and Bylaws at the first meeting of stockholders called for that purpose or any adjournment thereof;
- (d) by the Company, if any of the conditions in Section 7.2 becomes impossible to fulfill, other than for reasons totally within the control of the Company, and shall not have been waived in writing by the Company;
- (e) subject to compliance in all respects with Section 5.3(f), by the Company at any time before approval of this Agreement and the Merger by the stockholders of the Company, concurrently with the recommendation or approval by the Company Board of a Superior Proposal, but only if:
 - (i) the Company Board shall have determined in good faith in the exercise of its fiduciary duties (after consultation with the Company's financial adviser and outside legal counsel) to recommend or approve the Superior Proposal;
 - (ii) at all times before such termination the Company complied with Section 5.3;
 - (iii) before or simultaneously with such termination, the payments provided for in Section 8.3 are made to Parent; and
 - (iv) substantially concurrently with such termination, the Company enters into a definitive agreement with respect to the Superior Proposal; or
 - (f) by Parent, if:
 - (i) (A) the Company fails to call or hold the stockholders' meeting contemplated by Section 5.2(a), to solicit proxies in connection with such meeting in favor of adoption and approval of this Agreement and the Merger, or to conduct the vote to consider this Agreement and the Merger at the meeting or any adjournment thereof, each in compliance in all respects with Section 5.2; (B) the Company Board fails to recommend the Merger to the Company's stockholders or makes, or publicly announces an intent to make, an Adverse Recommendation Change; (C) the Company Board or any committee thereof accepts, recommends, or approves, or makes a determination to accept, recommend, or approve, an

Acquisition Proposal or a proposal for a Third-Party Transaction; or (D) the Company enters into any agreement for a Third-Party Transaction; or

- (ii) the Company materially breaches Section 5.3 of this Agreement.
- **8.2** <u>Procedure and Effect of Termination</u>. In the event of termination and abandonment of the Merger by the Company or Parent under Section 8.1, written notice thereof shall forthwith be given to the other parties hereto and this Agreement shall terminate and the Merger shall be abandoned without further action by any of the parties hereto. If this Agreement is terminated as provided herein, no party hereto shall have any liability or further obligation to any other party to this Agreement except as stated in Section 8.3 and Article IX or except with respect to a material breach of this Agreement by a party hereto, in which event, if the other party terminates this Agreement, then the terminating party may take any action or pursue any remedy available to it under Applicable Laws.

8.3 Termination Fee.

- (a) If this Agreement is terminated under Section 8.1 and Parent is entitled to a Termination Fee (as defined in Section 8.3(b)) under Section 8.3(c), the Company shall, at the same time payment of the Termination Fee is required to be made in the event of a termination under Section 8.1(e) or 8.1(f) or, if the Transaction Expenses have not been previously paid pursuant to this Section 8.3(a), at the same time payment of the Termination Fee is required to be made under Section 8.3(c), pay Parent, in immediately available funds, an amount equal to all documented reasonable out-of-pocket expenses incurred by or on behalf of Parent, Operating Sub, or Sub in connection with its due diligence investigation of the Company and the negotiation, preparation, financing, and execution of this Agreement and the transactions contemplated hereby, including financial-advisory, legal, accounting, and other reasonable fees and expenses ("<u>Transaction Expenses</u>"), provided that the aggregate amount payable by the Company to Parent under this Section 8.3(a) will not exceed C\$500,000.
- (b) If this Agreement is terminated under Section 8.1(e) or 8.1(f) (provided that in the case of a termination by Parent under Section 8.1(f)(i)(B), such termination is made within five business days of the event giving rise to such termination right), then the Company shall, before or simultaneously with such termination, pay Parent by wire transfer a fee in immediately available funds to an account designated by Parent (a "<u>Termination Fee</u>") of C\$4.3 million.
- (c) If (i) this Agreement is terminated by the Company or Parent under Section 8.1(b)(i), by Parent under Section 8.1(c)(ii), or by the Company or Parent under Section 8.1(c)(ii), and (ii) before such termination (A) any person or group shall have informed the Company (or the Company Board or any officer of the Company) after the date hereof that such person or group proposes, intends to propose, is considering proposing, or will or may, if the Merger is delayed, abandoned, or not approved by the Company's stockholders, propose, a Third-Party Transaction, or (B) any such person or

group or the Company publicly announces or it shall otherwise have become publicly known that such person or group has proposed, intends to propose, is considering proposing, or will or may, if the Merger is delayed, abandoned, or not approved by the Company's stockholders, propose, a Third-Party Transaction, and (iii) within one year after such termination the Company or any of its subsidiaries enters into a definitive agreement for, or consummates, a Third-Party Transaction (whether or not involving such person or group), then the Company shall, substantially concurrently with the time it or any of its subsidiaries enters into such definitive agreement for, or consummates, such Third Party Transaction, pay to Parent the Termination Fee.

(d) In no event will more than one Termination Fee be payable under this Section 8.3.

Article IX Miscellaneous

- 9.1 Termination of Representations and Warranties. The respective representations and warranties of the parties hereto will not survive the Effective Time.
- 9.2 Amendment and Modification. To the extent permitted by Applicable Law, this Agreement may be amended, modified, or supplemented by written agreement of the Company, Parent, Operating Sub, and Sub at any time before the Effective Time with respect to any of the terms contained herein, except that, after the meeting of stockholders contemplated by Section 5.2(a), there shall not be made any amendment that by Applicable Law requires further approval by the stockholders of the Company or Parent without such further approval of such stockholders.
- **9.3** Waiver of Compliance; Consents. Any failure of Parent, Operating Sub, or Sub, on the one hand, or the Company, on the other hand, to comply with any obligation, covenant, agreement, or condition herein may be waived in writing by Parent or the Company, respectively, but such waiver or failure to insist upon strict compliance with such obligation, covenant, agreement, or condition shall not operate as a waiver of, or estoppel with respect to, any subsequent or other failure. Whenever this Agreement requires or permits consent by or on behalf of any party hereto, such consent shall be given in writing in a manner consistent with the requirements for a waiver of compliance as set forth in this Section 9.3.
- 9.4 Expenses. Except as otherwise provided in Section 8.3, all expenses incurred in connection with this Agreement and the consummation of the transactions contemplated hereby shall be paid by the party incurring such expenses.
- 9.5 Additional Agreements. In case at any time after the Effective Time any further action is necessary or desirable to carry out the purposes of this Agreement, the proper officers and directors of each corporation that is a party to this Agreement shall take all such necessary action.

9.6 Notices. All notices and other communications hereunder shall be in writing and shall be deemed given if delivered personally, effective when delivered, or by express courier service, effective one business day after delivery to such courier, or by telecopy, effective when transmitted and a confirmation is received, provided the same is on a business day, and, if not, on the next business day, to the parties at the following addresses (or at such other address for a party or to such other person's attention as shall be specified by like notice):

(a) If to Parent, Operating Sub, or Sub:

Deluxe Corporation 3680 Victoria St. N. Shoreview, Minnesota 55126 Attention: Chief Executive Officer Fax No.: (651) 787-2749

with a copy to:

Deluxe Corporation 3680 Victoria St. N. Shoreview, Minnesota 55126 Attention: General Counsel Fax No.: (651) 787-2749

and

Faegre & Benson LLP 2200 Wells Fargo Center 90 South Seventh Street Minneapolis, Minnesota 55402 Attention: Michael A. Stanchfield Fax No.: (612) 766-1600

(b) If to the Company:

Hostopia.com Inc. 5915 Airport Road Suite 1100 Mississauga, Ontario, Canada L4V 1T1 Attention: Chief Financial Officer Fax No.: (905) 673-1551 with a copy to:

Pillsbury Winthrop Shaw Pittman LLP 2475 Hanover Street Palo Alto, California 94304 Attention: Jeffrey S. Harrell Fax No.: (650) 223-4545

9.7 <u>Assignment; No Third-Party Beneficiaries</u>. This Agreement will be binding upon and will inure to the benefit of the parties hereto and their respective successors and permitted assigns, but neither this Agreement nor any of the rights, interests, or obligations hereunder may be assigned by any party hereto without the prior written consent of the other parties (except that Sub may assign to any other direct or indirect wholly-owned subsidiary of Parent any and all rights and obligations of Sub under this Agreement, provided that any such assignment will not relieve Parent from any of its obligations under this Agreement). Except as expressly set forth in Sections 6.1 and 6.2 with respect to Indemnified Parties, this Agreement is not intended to confer upon any person except the parties hereto any rights or remedies hereunder.

9.8 Interpretation.

- (a) In this Agreement, unless the context clearly requires otherwise:
 - (i) words of any gender include each other gender;
 - (ii) words using the singular or plural number also include the plural or singular number, respectively;
- (iii) "person" means an individual, a partnership, a joint venture, a corporation, a limited liability company, a trust, an incorporated organization, and a Governmental Authority;
 - (iv) "affiliate" has the meaning set forth in Rule 12b-2 promulgated under the Exchange Act;
- (v) "subsidiary" of any specified corporation means any corporation or other entity of which the outstanding securities having ordinary voting power to elect a majority of the board of directors or other persons performing similar functions are directly or indirectly owned by such specified corporation, and "wholly-owned subsidiary" of any specified corporation means any subsidiary of the corporation all of the outstanding capital stock or other equity securities of which are directly or indirectly owned by the corporation;
- (vi) "business day" means any day other than a Saturday, Sunday, or a day that is a statutory holiday under the laws of Canada, the United States of America, the Province of Ontario, or the State of Minnesota;

- (vii) the terms "Article" or "Section" refer to the specified Article or Section of this Agreement;
- (viii) "including" (or any variation thereof) means "including without limitation";
- (ix) a reference to a number of days refers to calendar days unless otherwise specified;
- (x) all accounting terms used, but not expressly defined, have the meanings given to them under GAAP;
- (xi) "C\$" means the lawful currency of Canada, and "US\$" means the lawful currency of the United States of America;
- (xii) the words "herein," "hereof," "hereunder," and similar words refer to this Agreement as a whole and not to any particular Article, Section, or subsection of this Agreement;
 - (xiii) all references to statutes or regulations are deemed to refer to those statutes and regulations as amended from time to time; and
- (xiv) when calculating the period of time before which, within which, or following which any act is to be done or step taken pursuant to this Agreement, the date that is the reference date in calculating such period shall be excluded, and, if the last day of such period is not a business day, then the period in question shall end on the next business day.
- (b) The parties have participated jointly in the negotiation and drafting of this Agreement and, if an ambiguity or question of intent or interpretation arises, it is the intent of the parties that this Agreement shall be construed as jointly drafted by the parties and that no presumption or burden of proof shall arise favoring or disfavoring any party by virtue of the authorship of any provision of this Agreement.
- (c) For purposes of all conversions from US\$ to C\$ or from C\$ to US\$ required to effect the terms of this Agreement, the exchange rate used for such conversions shall be calculated by reference to the CAD (composite) moving average of the spot rate on the date on which the Effective Time occurs, as quoted on the Bloomberg system.
- 9.9 Governing Law. This Agreement shall be governed by the laws of the State of Delaware, without regard to its conflict-of-laws rules. The parties hereto expressly consent to jurisdiction and venue for any litigation arising out of this Agreement in the Chancery Court in the State of Delaware.

- 9.10 Counterparts. This Agreement may be executed in two or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.
- 9.11 Headings. The article and section headings contained in this Agreement are solely for the purpose of reference, and are not part of the agreement of the parties and shall not affect in any way the meaning or interpretation of this Agreement.
- 9.12 <u>Disclosure Schedule</u>. Matters reflected in the Disclosure Schedule are not necessarily limited to matters required by this Agreement to be reflected therein. Such additional matters are set forth for informational purposes and do not necessarily include other matters of a similar nature that are not required to be reflected therein. Matters disclosed pursuant to any Section of the Disclosure Schedule will be deemed to be disclosed with respect to all Sections of the Disclosure Schedule only to the extent this Agreement requires such disclosure and only to the extent such disclosure reasonably relates to such Section of the Disclosure Schedule. Nothing in the Disclosure Schedule shall be deemed adequate to disclose any matter, including an exception to a representation or warranty or covenant made herein, however, unless the Disclosure Schedule identifies the matter in a manner reasonably sufficient to inform Parent of the nature of the matter.
- **9.13** Specific Performance. Each of the parties hereto acknowledges and agrees that the other party would be damaged irreparably in the event any of the provisions of this Agreement are not performed in accordance with their specific terms or otherwise are breached. Accordingly, each of the parties hereto agrees that the other party shall be entitled to an injunction to prevent breaches of the provisions of this Agreement and to enforce specifically this Agreement and the terms and provisions hereof in addition to any other remedy to which they may be entitled.
- **9.14** Entire Agreement. This Agreement, including the Disclosure Schedule, the exhibits hereto, and the documents and instruments referred to herein, together with the Confidentiality Agreement, embody the entire agreement and understanding of the parties hereto in respect of the subject matter contained herein. This Agreement supersedes all prior agreements and understandings among the parties with respect to such subject matter, except for the Confidentiality Agreement.

[signature page follows]

The parties hereto have caused this Agreement to be signed by their respective duly authorized officers on the date first written above.

DELUXE CORPORATION

By: /s/ Lee J. Schram

Lee J. Schram

Chief Executive Officer

DELUXE BUSINESS OPERATIONS, INC.

By: /s/ Lee J. Schram

Lee J. Schram President

HELIX MERGER CORP.

By: /s/ Lee J. Schram

Lee J. Schram President

HOSTOPIA.COM INC.

By: /s/ Colin Campbell

Colin Campbell Chief Executive Officer

EXHIBIT A

Certificate of Incorporation of Hostopia.com Inc.

Article I Name

The name of the Corporation is Hostopia.com Inc. (hereinafter referred to as the "Corporation").

Article II Registered Agent

The address of the Corporation's registered office in the State of Delaware is c/o Corporation Trust Center, 1209 Orange Street in the City of Wilmington, County of New Castle, Delaware 19801. The name of the Corporation's registered agent for service of process at such address is The Corporation Trust Company.

Article III Purpose

The purpose of the Corporation is to engage in any lawful act or activity for which corporations may be organized under the Delaware General Corporation Law (the "DGCL").

Article IV Capital Stock

The total number of shares of stock that the Corporation has the authority to issue is 1,000 shares, all of which shall be designated Common Stock, par value \$0.001 per share.

Article V Bylaws

In furtherance and not in limitation of the powers conferred by statute, the Board of Directors is expressly authorized to make, repeal, alter, amend and rescind any or all of the Bylaws of the Corporation.

Article VI Number of Directors

The number of directors of the Corporation shall be determined in the manner set forth in the Bylaws of the Corporation.

Article VII Election of Directors

Elections of directors need not be by written ballot unless the Bylaws of the Corporation shall so provide.

Article VIII

Indemnification; Limitation on Director's Liability

- (a) To the fullest extent permitted by applicable law, this Corporation is authorized to provide indemnification of (and advancement of expenses to) directors, officers, employees and agents (and any other persons to which the DGCL permits this Corporation to provide indemnification) through Bylaw provisions, agreements with such agents or other persons, vote of stockholders or disinterested directors or otherwise, in excess of the indemnification and advancement otherwise permitted by Section 145 of the DGCL, subject only to limits created by the DGCL (statutory or non-statutory), with respect to action for breach of duty to the Corporation, its stockholders, and others.
- (b) Each person who was or is made a party or is threatened to be made a party to or is in any way involved in any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative (hereinafter a "proceeding"), including any appeal therefrom, by reason of the fact that he or she, or a person of whom he or she is the legal representative, is or was a director or officer of the Corporation or a direct or indirect subsidiary of the Corporation, or is or was serving at the request of the Corporation as a director or officer of another entity or enterprise, or was a director or officer of a foreign or domestic corporation which was a predecessor corporation of the Corporation or of another entity or enterprise at the request of such predecessor corporation, shall be indemnified and held harmless by the Corporation, and the Corporation shall advance all expenses incurred by any such person in defense of any such proceeding prior to its final determination, to the fullest extent authorized by the DGCL. In any proceeding against the Corporation to enforce these rights, such person shall be presumed to be entitled to indemnification and the Corporation shall have the burden of proving that such person has not met the standards of conduct for permissible indemnification set forth in the DGCL. The rights to indemnification and advancement of expenses conferred by this Article VIII shall be presumed to have been relied upon by the directors and officers of the Corporation in serving or continuing to serve the Corporation and shall be enforceable as contract rights. Said rights shall not be exclusive of any other rights to which those seeking indemnification may otherwise be entitled. The Corporation may, upon written demand presented by a director or officer of the Corporation or of a direct or indirect subsidiary of the Corporation, or by a person serving at the request of the Corporation as a director or officer of another entity or enterprise, enter into contracts to p

(c) If a claim under this Article VIII is not paid in full by the Corporation within sixty (60) days after a written claim has been received by the Corporation, the claimant may at any time thereafter bring suit against the Corporation to recover the unpaid amount of the

claim and, if successful in whole or in part, the claimant shall be entitled to be paid also the expenses of prosecuting such claim. It shall be a defense to any such action (other than an action brought to enforce the right to be advanced expenses incurred in defending any proceeding prior to its final disposition where the required undertaking, if any, has been tendered to the Corporation) that the claimant has not met the standards of conduct which make it permissible under the DGCL for the Corporation to indemnify the claimant for the amount claimed, but the claimant shall be presumed to be entitled to indemnification and the Corporation shall have the burden of proving that the claimant has not met the standards of conduct for permissible indemnification set forth in the DGCL.

- (d) If the DGCL is hereafter amended to permit the Corporation to provide broader indemnification rights than said law permitted the Corporation to provide prior to such amendment, the indemnification rights conferred by this Article VIII shall be broadened to the fullest extent permitted by the DGCL, as so amended.
- (e) No director of the Corporation shall be personally liable to the Corporation or any stockholder for monetary damages for breach of fiduciary duty as a director, except for any matter in respect of which such director shall be liable under Section 174 of the DGCL or any amendment thereto or shall be liable by reason that, in addition to any and all other requirements for such liability, such director (1) shall have breached the director's duty of loyalty to the Corporation or its stockholders, (2) shall have acted in a manner involving intentional misconduct or a knowing violation of law or, in failing to act, shall have acted in a manner involving intentional misconduct or a knowing violation of law, or (3) shall have derived an improper personal benefit. If the DGCL is hereafter amended to authorize the further elimination or limitation of the liability of a director, the liability of a director of the Corporation shall be eliminated or limited to the fullest extent permitted by the DGCL, as so amended.
- (f) Any amendment, repeal or modification of the foregoing provisions of this Article VIII shall not adversely affect any right or protection of any director, officer or other agent of the Corporation existing at the time of, or increase the liability of any director of the Corporation with respect to any acts or omissions of such director, officer or other agent occurring prior to, such amendment, repeal or modification.

Article IX Amendment of Certificate of Incorporation

The Corporation reserves the right to amend, alter, change or repeal any provision contained in this Certificate of Incorporation, in the manner now or hereafter prescribed by statute, and all rights conferred upon stockholders, directors, or any other persons herein are granted subject to this reservation.

VOTING AGREEMENT

This Voting Agreement is dated as of June 18, 2008, among Deluxe Corporation, a Minnesota corporation ("<u>Parent</u>"), and the persons listed on <u>Annex A</u> hereto, each of whom is a stockholder (a "<u>Stockholder</u>") of Hostopia.com Inc., a Delaware corporation (the "<u>Company</u>").

Recitals

- A. Parent, Deluxe Business Operations, Inc., a Delaware corporation and wholly owned subsidiary of Parent ("Operating Sub"), Helix Merger Corp., a Delaware corporation and wholly owned subsidiary of Operating Sub ("Sub"), and the Company are entering into an Agreement and Plan of Merger dated as of the date hereof (the "Merger Agreement"), pursuant to which Sub will merge (the "Merger") with and into the Company and the existing stockholders of the Company will receive cash from Parent in exchange for their shares of capital stock of the Company ("Company Capital Stock").
 - B. The execution and delivery of this Agreement is a condition precedent to Parent, Operating Sub and Sub entering into the Merger Agreement.
 - C. Capitalized terms used but not otherwise defined herein have the meanings ascribed to them in the Merger Agreement.

Therefore, the parties hereby agree as follows:

1. Voting; Proxy.

- (a) During the term of this Agreement, at each meeting of the Company's stockholders convened, and at any adjournment or postponement thereof, to consider and vote upon the approval of the Merger Agreement and the Merger (or as a part of any such action by written consent of the stockholders in lieu of a meeting), each Stockholder agrees to vote all shares of Company Capital Stock owned of record by such Stockholder as of the record date for the vote or written consent (including, any shares of Company Capital Stock over which the Stockholder has voting power results from the Stockholder having been named as proxy pursuant to any proxy solicitation conducted by the Company's board of directors) (i) in favor of the approval and adoption of the Merger Agreement and the Merger; (ii) against approval of any proposal made in opposition to or in competition with the consummation of the Merger and the Merger Agreement; and (iii) against any other proposal or action that would, or would reasonably be expected to, prohibit or discourage the Merger.
 - (b) Each Stockholder has executed and delivered to Parent an irrevocable proxy in the form of Annex B.
- (c) Each Stockholder will be present, in person or by proxy, at all meetings of stockholders of the Company so that all shares of Company Capital Stock are counted for the purposes of determining the presence of a quorum at such meetings.

(d) This Agreement is intended to bind each Stockholder only with respect to the specific matters set forth herein, and will not prohibit a Stockholder from acting in accordance with such Stockholder's fiduciary duties as an officer or director of the Company.

2. No Solicitation; No Transfer.

- (a) Until the earlier of the Effective Time or the termination of this Agreement pursuant to its terms, no Stockholder will, nor will any Stockholder authorize or permit any of the Stockholder's representatives to, and each Stockholder will instruct each such representative not to, directly or indirectly, take any actions prohibited by Section 5.3 of the Merger Agreement, as if such Stockholder were a party thereto.
- (b) During the term of this Agreement, none of the Stockholders may sell, pledge, assign, or otherwise transfer, or authorize, propose, or agree to the sale, pledge, assignment, or other transfer of, any of such Stockholder's shares of Company Capital Stock, unless (a) at least two business days' written notice of the proposed transfer is provided to Parent and (b) the intended transferee agrees in writing to be bound by this Agreement as if such person were such Stockholder.
- 3. Representations and Warranties. Each Stockholder represents and warrants to Parent as follows:
- (a) *Authority*. The Stockholder has the requisite power and authority to enter into this Agreement, to perform the Stockholder's obligations hereunder, and to consummate the transactions contemplated hereby. No other proceedings or actions on the part of the Stockholder are necessary to authorize the execution, delivery, or performance of this Agreement or the consummation of the transactions contemplated hereby. This Agreement has been duly executed and delivered by the Stockholder and constitutes a valid and binding obligation, enforceable against the Stockholder in accordance with its terms, except as the enforceability hereof may be limited by the Enforcement Limitations
- (b) *Title; Authority to Vote Shares*. The Stockholder is the sole owner of record and has sole voting power over the number of shares of Company Capital Stock set forth opposite the Stockholder's name on Annex A, and such shares are held by the Stockholder free and clear of all Encumbrances that would limit or otherwise affect the Stockholder's voting rights with respect to the Company Capital Stock set forth on Annex A or otherwise prevent the Stockholder from performing the Stockholder's obligations hereunder.
- (c) Noncontravention. Neither the Stockholder's execution and delivery of this Agreement, nor the Stockholder's consummation of any of the transactions contemplated hereby, nor the Stockholder's compliance with any of the provisions hereof, will violate, conflict with, or result in a breach of, or constitute a default (or an event that, with notice or lapse of time or both, would constitute a default) under, or result in the creation of any Encumbrance upon any of the Stockholder's shares of Company Capital Stock under, any

agreement or instrument to which the Stockholder is a party or any statute, rule, regulation, judgment, order, decree, or other legal requirement applicable to the Stockholder.

- (d) Litigation. (i) There is no claim, action, proceeding, or investigation pending or, to the Stockholder's knowledge, threatened against or relating to the Stockholder before any Governmental Authority, and (ii) the Stockholder is not subject to any outstanding order, writ, injunction, or decree, that, in the case of clause (i) or (ii), if determined adversely, would limit or otherwise affect the Stockholder's voting rights with respect to the Company Capital Stock set forth on Annex A or otherwise prohibit the Stockholder from performing the Stockholder's obligations hereunder.
- (e) Reliance. Each Stockholder understands and acknowledges that Parent is entering into the Merger Agreement in reliance upon such Stockholder's execution, delivery, and performance of this Agreement. Each Stockholder acknowledges that its irrevocable proxy is granted in consideration of the execution and delivery of the Merger Agreement by Parent.
- 4. <u>Termination</u>. This Agreement shall terminate automatically and without further action on behalf of any party at the earlier of (a) the Effective Time or (b) the date and time, if any, that the Merger Agreement is terminated pursuant to its terms. In the event of a termination of this Agreement pursuant to this Section 4, this Agreement shall forthwith become void and there shall be no liability or obligation on the part of any party, but nothing herein shall release the Stockholder for any breach of this Agreement. If this Agreement is terminated, the proxy of the Stockholder delivered under Section 1(b) hereof shall also terminate and be of no further force or effect, and Parent shall promptly return the proxy to the Stockholder.

5. Miscellaneous.

(a) *Notices*. All notices and other communications hereunder shall be in writing and shall be deemed given if delivered personally, effective when delivered, or by express courier service, effective one business day after delivery to such courier, or by telecopy, effective when transmitted and a confirmation is received, provided the same is on a business day, and, if not, on the next business day, to the parties at the following addresses (or at such other address for a party or to such other person's attention as shall be specified by like notice):

If to a Stockholder: to the address listed for the Stockholder on Annex A.

If to Parent, to:

Deluxe Corporation 3680 Victoria St. N. Shoreview, MN 55126 Facsimile: (651) 787-2749 Attention: General Counsel

- (b) *Headings*. The headings contained in this Agreement are solely for the purpose of reference, and are not part of the agreement of the parties and shall not affect in any way the meaning or interpretation of this Agreement.
- (c) Counterparts. This Agreement may be executed in two or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.
- (d) Entire Agreement. This Agreement (along with the documents and instruments referred to herein, including the Merger Agreement), embody the entire agreement and understanding of the parties hereto in respect of the subject matter contained herein. This Agreement supersedes all prior agreements and understandings among the parties with respect to such subject matter.
- (e) Severability. The invalidity or unenforceability of any provision of this Agreement shall not affect the validity or enforceability of any other provisions of this Agreement. Whenever possible, each provision of this Agreement will be interpreted in such manner as to be effective and valid under Applicable Law.
- (f) Governing Law. This Agreement shall be governed by the laws of the State of Delaware, without regard to its conflict-of-laws rules. The parties hereto expressly consent to jurisdiction and venue for any litigation arising out of this Agreement in the Chancery Court in the State of Delaware.
- (g) Assignment. Neither this Agreement nor any of the rights, interests, or obligations hereunder may be assigned by any party, whether by operation of law or otherwise, without the express written consent of the other party. Subject to the preceding sentence, this Agreement will be binding upon, inure to the benefit of, and be enforceable by the parties and their respective successors, heirs, legal representatives, and permitted assigns. The representations, agreements, and obligations of the Stockholder contained herein shall survive the death or incapacity of the Stockholder and shall be binding upon the heirs, personal representatives, successors, and assigns of the Stockholder.
- (h) *Remedies*. In addition to all other remedies available, the parties agree that, in the event of a breach by a party of any of its obligations hereunder, the non-breaching party shall be entitled to specific performance or injunctive relief.
 - (i) Defined Terms. All capitalized terms used but not defined herein have the meanings given them in the Merger Agreement.
- (j) Waiver. Prior to the Closing Date, any provision of this Agreement may be waived by the party or parties benefited by the provision. Waiver of any term or condition of this Agreement (including any extension of time required for performance) will only be effective if in writing and will not be construed as a waiver of any subsequent breach or waiver of the same term or condition, or a waiver of any other term or condition of this

Agreement. No delay on the part of any party in exercising any right, power or privilege hereunder will operate as a waiver thereof.

(k) Further Assurances. From time to time, at Parent's request and without further consideration, each Stockholder shall execute and deliver such additional documents and take all such further lawful action as may be reasonably necessary to consummate and make effective, in the most expeditious manner practicable, the transactions contemplated by this Agreement.

[Signature Page Follows]

Each of the parties has signed this Agreement as of the date first written above.

DELUXE CORPORATION

By: /s/ Lee J. Schram

Lee J. Schram Chief Executive Officer

[Signature Page to Voting Agreement]

STOCKHOLDERS

/s/ William Campbell By: William Campbell By: /s/ Colin Campbell Colin Campbell /s/ Michael Cytrynbaum Michael Cytrynbaum By: /s/ Matthew George Mathew George /s/ Robert Kidd By: Robert Kidd By: /s/ Randall Bast Randall Bast /s/ Christopher Scatliff Christopher Scatliff

[Signature Page to Voting Agreement]

STOCKHOLDERS (continued)

1999 WILLIAM CAMPBELL FAMILY TRUST

By: /s/ William Campbell

Name: William Campbell

Title: Trustee

1555706 ONTARIO LIMITED

/s/ William Campbell

By: /s/ Colin Campbell
Name: William Campbell and Colin Campbell

Title: Directors

TELUS CORPORATION

By: /s/ Robert Gardner

Name: Robert Gardner

Title: Senior Vice President & Treasurer

[Signature Page to Voting Agreement]

Annex A

Name and Address	Number of Shares of Common Stock
William Campbell*	869,806
Colin Campbell*	1,154,006
Michael Cytrynbaum*	0
Mathew George*	0
Robert Kidd*	40,000
Randall Bast*	25,000
Christopher Scatliff*	83,100
1999 William Campbell Family Trust*	264,000
1555706 Ontario Limited*	851,063
TELUS Corporation 12 — 3777 Kingsway Burnaby, B.C. Canada V5H 3Z7	851,063

^{*} Address is: c/o Hostopia.com Inc., 5915 Airport Road, 11th Floor, Mississauga, Ontario, Canada L4V 1T1.

IRREVOCABLE PROXY

The undersigned, revoking all prior proxies, hereby constitutes and appoints each of Lee J. Schram and Anthony C. Scarfone the true and lawful attorney, with full power of substitution, for and in the name of the undersigned to vote, at any time before the Termination, all shares of capital stock of Hostopia.com Inc., a Delaware corporation (the "Company"), or other shares of capital stock of the Company entitled to vote on the business to be transacted, (a) registered in the name of the undersigned at the record date for such vote or written consent, or (b) except as set forth below, over which the undersigned has voting power by power of attorney or other contractual arrangements with the owner of record (collectively, the "Shares"), at any meeting of the stockholders of the Company, and at all adjournments thereof, and pursuant to any consent of the stockholders in lieu of a meeting or otherwise, (i) in favor of the approval and adoption of the Merger Agreement and the Merger; (ii) against approval of any proposal made in opposition to or in competition with the consummation of the Merger and the Merger Agreement; and (iii) against any other proposal or action that would, or would reasonably be expected to, prohibit or discourage the Merger.

This Proxy is given with respect to the approval of the Agreement and Plan of Merger among Deluxe Corporation, a Minnesota corporation ("<u>Parent</u>"), Deluxe Business Operations, Inc., a Delaware corporation and wholly owned subsidiary of Parent ("<u>Operating Sub</u>"), Helix Merger Corp., a Delaware corporation and wholly owned subsidiary of Operating Sub, and the Company, dated as of June 18, 2008 (the "<u>Merger Agreement</u>") and the Merger contemplated thereunder (the "<u>Merger"</u>"). This Proxy is given to induce Parent to enter into the Merger Agreement, is coupled with an interest, and is irrevocable; provided, however, that this Proxy shall terminate automatically and without further action on behalf of the undersigned upon the termination of the Voting Agreement, dated as of the date hereof, between Parent and the undersigned (the "<u>Termination</u>").

Notwithstanding clause (b) of the first paragraph above, this Proxy shall not include any shares of capital stock of the Company that are not subject to clause (a) of the first paragraph above for which the undersigned's only voting power results from the undersigned having been named as proxy pursuant to any proxy solicitation conducted by the Company's board of directors in connection with a meeting of the Stockholders of the Company and over which the undersigned does not otherwise have voting power with respect thereto.

	[Name]	
The undersigned has executed this Proxy as of this 18th day of June, 2008.		
The undersigned neleby fathres and commiss an that the proxies named netern may fawfuny do of cause to be done by virtue nelebi.		





NEWS RELEASE

June 19, 2008

Deluxe Corporation

P.O. Box 64235 St. Paul, MN 55164-0235 (651) 483-7111

For additional information:

Terry D. Peterson

Deluxe Corporation VP, Investor Relations and Chief Accounting Officer (651) 787-1068

Paul D. Engels

Hostopia.com Inc. VP and Chief Marketing Officer (416) 883-6706

Deluxe to Acquire Hostopia.com

Acquisition Provides Web-Hosting Capabilities for Small Businesses

St. Paul, Minn. and Mississauga, Ont. – Deluxe Corporation (NYSE: DLX) and Hostopia.com Inc. (TSX: H) today announced that they have entered into a definitive agreement for Deluxe to acquire Hostopia, a leading provider of web services that enable small and medium-sized businesses to establish and maintain an Internet presence. Deluxe will acquire Hostopia in an all-cash transaction that values Hostopia at approximately C\$124 million. Under terms of the Merger Agreement, Deluxe will pay C\$10.55 in cash for each outstanding Hostopia common share. Subject to a final accounting allocation of the purchase price, the acquisition is not expected to have a significant impact to Deluxe's earnings and operating cash flow for fiscal 2008.

"We are very excited with the prospect of providing Hostopia's industry-leading, private-label web services to our small business customers and collectively growing our product and service offerings as we move forward together," said Lee Schram, Chief Executive Officer of Deluxe. "Hostopia has grown top line revenue in excess of 20 percent in each of the last three years. While we have substantial knowledge and expertise in selling to the small business market, we believe that Hostopia's ability to attract new customers and provide additional services to that market will increase the power of our offerings. This acquisition is in line with our long-term strategy of broadening the products and services we offer to small businesses to manage, promote and grow their businesses. We also believe that the transaction will offer significant value to our shareholders as we continue to introduce growing products and services with recurring revenue streams."

Colin Campbell, Chief Executive Officer of Hostopia, said, "After careful consideration, the Hostopia Board of Directors has unanimously approved this transaction with Deluxe. We believe the transaction delivers outstanding value to the Hostopia stockholders. Deluxe and Hostopia have many similarities, including a desire to help small businesses grow and promote their businesses. Our customers will benefit from augmented product and service offerings

1

which include small business logo design, print-on-demand services and e-mail marketing, all of which have demonstrated demand from small businesses. In addition, Deluxe brings access to new channels for Hostopia including direct mail to small businesses and sales through financial institutions. We look forward to providing web-hosting capabilities to small businesses as a significant component of the services that Deluxe offers to those businesses."

The transaction, which has been unanimously approved by the Boards of Directors of both companies, is structured as a merger under applicable law. It will require the approval of Hostopia stockholders holding a majority of the outstanding Hostopia common shares at a special meeting to be called to consider the transaction. Hostopia stockholders holding in aggregate approximately 35% of the issued and outstanding shares of Hostopia common stock have agreed to vote their Hostopia common shares in favor of the transaction as long as the merger agreement is in effect, pursuant to a voting agreement with Deluxe. The transaction is also subject to certain other customary closing conditions. The transaction is expected to close in the third quarter of 2008.

About Deluxe Corporation

Deluxe Corporation, through its industry-leading businesses and brands, helps financial institutions and small businesses better manage, promote, and grow their businesses. Deluxe uses direct marketing, distributors, and a North American sales force to provide a wide range of customized products and services: personalized printed items (checks, forms, business cards, stationery, greeting cards, labels, and retail packaging supplies), promotional products and merchandising materials, fraud prevention services, and customer retention programs. Deluxe also sells personalized checks and accessories directly to consumers. For more information about Deluxe Corporation, visit www.Deluxe.com.

About Hostopia.com Inc.

Hostopia.com Inc. is a leading provider of web services that enable small and medium-sized businesses to establish and maintain an Internet presence. Hostopia's customers are communication services providers, including telecommunication carriers, cable companies, internet service providers, domain registrars, and web hosting service providers. Hostopia's customers purchase its web services on a wholesale basis and resell these services under their own brands to small and medium-sized businesses. Hostopia provides customers with the technology, infrastructure, and support services to enable them to offer web services, while saving them research and development as well as capital and operating costs typically associated with the design, development, and delivery of web services.

Forward-Looking Statements

Certain statements contained in this release may be deemed to be forward-looking statements under certain securities laws, including the "safe harbor" provisions of the United States Private Securities Litigation Reform Act of 1995 and in any applicable Canadian securities legislation, and Deluxe Corporation and Hostopia.com Inc. intend that such forward-looking statements be subject to the safe-harbor created thereby. These forward-looking statements include, but are not

limited to, statements with respect to the acquisition of Hostopia by Deluxe and statements concerning Deluxe's and Hostopia's, or their management's, current intentions, expectations, beliefs, projections or predictions about future results or events. Forward-looking statements are typically identified by words such as "believe." "expect." "forecast," "anticipate," "intend," "estimate," "plan" and "project" and similar expressions of future or conditional verbs such as "will," "may," "should," "could," or "would." By their very nature, forward-looking statements require Deluxe and Hostopia to make assumptions and are subject to inherent risks and uncertainties that are difficult to predict and are generally beyond the control of Deluxe and Hostopia, which give rise to the possibility that certain predictions, forecasts, projections, expectations and other forward-looking information, including statements about the acquisition of Hostopia by Deluxe, will not be achieved. Deluxe and Hostopia caution readers not to place undue reliance on these statements as a number of important factors could cause actual results or events to differ materially and adversely from the beliefs, plans, objectives, expectations, anticipations, estimates and intentions expressed in, or implied or projected by, the forward-looking statements. These factors include, but are not limited to, the following: the possibility that the acquisition of Hostopia by Deluxe does not close when expected or at all because required regulatory, shareholder or other approvals are not received or other conditions to the closing are not satisfied on a timely basis or at all; the risks and uncertainties associated with Deluxe's ability to complete the acquisition of Hostopia and to integrate Hostopia with Deluxe successfully; the ability to retain key personnel; the inherent unreliability of earnings, revenue and cash flow predictions due to numerous factors, many of which are beyond Deluxe's and Hostopia's control; developments in the demand for the combined companies' products and services; relationships with major customers and suppliers; unanticipated delays, costs and expenses inherent in the development and marketing of new products and services, including new ecommerce, customer loyalty and business services, and the failure of such new products and services to deliver the expected revenues and other financial targets; the impact of governmental laws and regulations; and competitive factors. Readers are cautioned that the foregoing list of important factors is not exhaustive. Additional information concerning these and other factors that could cause actual results and events to differ from Deluxe's and Hostopia's current expectations are contained in Deluxe's and Hostopia's public filings with the Securities and Exchange Commission, including but not limited to the factors discussed under "Risk Factors" in Deluxe Corporation's Form 10-O for the period ended March 31, 2008 and Hostopia's Form 10-O for the period ended December 31, 2007 (each as updated by subsequent filings with the Securities and Exchange Commission).

Except as required by law, Deluxe and Hostopia assume no obligation to update the forward-looking statements contained in this release to reflect events or circumstances after the date hereof or to reflect the occurrence of unanticipated events.

###